

# China's Role in Promoting Transboundary Resource Management in the Greater Mekong Basin (GMB)

# Malcolm F. McPherson Ash Center Senior Fellow in Development

March 2020





# China's Role in Promoting Transboundary Resource Management in the Greater Mekong Basin (GMB)

. . . . . . . . . .

Malcolm F. McPherson Ash Center Senior Fellow in Development

March 2020



## **ABOUT THE AUTHOR**

Malcolm McPherson, who earned his Ph.D. in economics from Harvard in 1980, is a senior research associate with the Vietnam Program at the Ash Center for Democratic Governance and Innovation. McPherson joined Harvard Kennedy School in 2000 after having worked for Harvard Institute for International Development (HIID) for 18 years. His research interests include international development, monetary policy, and exchange rate management, macroeconomic reform, and the relationships between education and economic growth. For HIID, he served as resident advisor in The Gambia (1985–89) and Zambia (1992–96). He co-edited with Steven Radelet a book on economic recovery in The Gambia (Harvard University Press, 1995) and, with Catharine Hill, a volume on economic reform in Zambia (Harvard University Press 2004). As a member of the Belfer Center, Center for Business and Government, and now the Ash Center, McPherson's research has dealt with public-private partnerships, capacity building in developing countries, post-conflict recovery, agricultural development, and poverty reduction. His most recent assignments include work in Georgia, Armenia, Liberia, Zambia, Timor Leste, Chile, Indonesia, and Myanmar.

## A NOTE FROM THE AUTHOR

This research is supported by a grant to the Ash Center for Democratic Governance and Innovation by the Harvard Global Institute, Office of the President, Harvard University. I have benefited from the research, perspectives, and opinions of scholars in the United States and Asia, and where possible, have acknowledged their contributions. I am grateful to Harvard's Anthony Saich and Dwight Perkins for detailed discussions on Chinese policy and institutions, and to Chayanis Krittasudthacheewa and Leonie Pearson of the Stockholm Environment Institute for criticisms and suggestions. Any errors are mine. CHINA'S ROLE IN PROMOTING TRANSBOUNDARY RESOURCE MANAGEMENT IN THE GREATER MEKONG BASIN (GMB)

# ABOUT THE ASH CENTER

The Roy and Lila Ash Center for Democratic Governance and Innovation advances excellence and innovation in governance and public policy through research, education, and public discussion. By training the very best leaders, developing powerful new ideas, and disseminating innovative solutions and institutional reforms, the Center's goal is to meet the profound challenges facing the world's citizens. The Ford Foundation is a founding donor of the Center. Additional information about the Ash Center is available at ash.harvard.edu.

This research paper is one in a series published by the Ash Center for Democratic Governance and Innovation at Harvard University's John F. Kennedy School of Government. The views expressed in the Ash Center Policy Briefs Series are those of the author(s) and do not necessarily reflect those of the John F. Kennedy School of Government or of Harvard University. The papers in this series are intended to elicit feedback and to encourage debate on important public policy challenges. This paper is copyrighted by the author(s). It cannot be reproduced or reused without permission. CHINA'S ROLE IN PROMOTING TRANSBOUNDARY RESOURCE MANAGEMENT IN THE GREATER

MEKONG BASIN (GMB)

# CONTENTS

| Section 1: Introduction   | 1  |
|---|----|
| Section 2: Background   | 4  |
| The Greater Mekong Basin  | 4  |
| Lancang-Mekong Cooperation  | 6  |
| The Administrative Inadequacy of Cooperation                      | 9  |
| Section 3: LMC Themes   | 12 |
| Connectivity  | 12 |
| Water Governance  | 17 |
| Overview  | 28 |
| Section 4: China's Role in Transboundary Resource Management?     | 28 |
| Section 5: Concluding Comments                                    | 32 |
| Annex I: The LMC and Public Policy                                | 34 |
| Annex II: Food Security, Irrigation, and Flood Control in Vietnam | 38 |
| References  | 40 |

# **SECTION 1: INTRODUCTION<sup>1</sup>**

This paper examines how China can improve transboundary resource management within the Greater Mekong Basin (GMB) through its participation in the Lancang-Mekong Cooperation (LMC).<sup>2</sup> Such improvement would ensure the efficient management and equitable development of the basin's natural resources and ecosystems.<sup>3</sup>

First proposed by Thailand in 2012, the LMC has been enthusiastically promoted by China since 2014. Formalized in the Sanya Declaration of March 2016,<sup>4</sup> it seeks deeper engagement between China and its GMB neighbors—Thailand, Laos or Lao PDR, Cambodia, Myanmar, and Vietnam.<sup>5</sup> The LMC's three cooperation pillars are "(1) political and security issues, (2) economic and sustainable development, and (3) social, cultural, and people-to-people exchanges."<sup>6</sup> The Declaration asserts that "practical cooperation" would "... start with five key priority areas ... connectivity, production capacity, cross-border economic cooperation, water resources, [and] agriculture and poverty reduction." Other measures, such as environmental protection and natural resource management, are also mentioned.<sup>7</sup> In November 2017, the LMC Secretariat was established in Beijing and by January 2018 the LMC Plan of Action (2018–2022) was published. This plan described the main topics that would be addressed and

.....

<sup>1</sup> This paper is a revision of a version prepared for a May 2019 workshop at Beijing Normal University. It incorporates criticisms and comments of workshop participants and takes into account what appear to be shifts in China's Belt and Road policies announced in Bejing in President Xi Jinping's speech to the Second Belt and Road Forum on April 26th.

<sup>2</sup> A reviewer criticized the idea that China should be "responsible for this," noting that transboundary resource management should be the joint responsibility of all countries. We agree but so far not all countries have exercised this joint responsibility, to the detriment of the GMB's environment. This paper seeks to understand how—and if—China could provide the necessary leadership.

<sup>3</sup> This sentence paraphrases how van der Linde et al. (2001), p. xvii defined "transboundary resource management."

<sup>4</sup> Sanya Declaration of the First Lancang-Mekong Cooperation (LMC) Leaders' Meeting, https://www.fmprc.gov .cn/mfa\_eng/wjdt\_665385/2649\_665393/t1350039.shtml.

<sup>5</sup> Xinhua 2016; Devlaeminck 2018.

<sup>6</sup> Sanya Declaration, para. 12

<sup>7</sup> The Sanya Declaration, art. 14, states "encourage sustainable and green development, enhance environmental protection and natural resources management, develop and utilize sustainably and efficiently clean energy resources, develop regional power market, and enhance exchange and transfer of clean energy technologies."

how the practical cooperation would evolve. It also outlined a sequence for the LMC's "foundation-laying" and consolidation activities.<sup>8</sup>

Formal communiques from high-level LMC meetings have affirmed that senior GMB officials welcome the potential for mutually beneficial cooperation. These communiques and other literature (primarily press coverage) highlight LMC activities. They include information on meetings of senior GMB officials, training sessions, LMC development projects, joint working groups on LMC themes, construction undertaken by Chinese state-owned enterprises (SOEs), planning seminars, exchange visits, workshops, and the creation of agencies to study water and the environment.

This is an impressive and vigorous start. The challenge, however, will be to maintain the momentum once the difficulties of cooperation as the LMC's key organizing principle become apparent. The initial emphasis on cooperation has been important in stimulating interest and eliciting commitments to engage from GMB leaders. All participants would wish to cooperate, particularly since China "made the call." Yet, over the medium and longer terms, all GMB countries—including China—will only remain engaged if LMC activities align with their goals, interests, capacities, and preferences. Accordingly, the LMC will have to shift its focus from cooperation to the *public policy* challenges and opportunities that will arise as GMB governments collectively formulate and implement the LMC agenda.

This point is critical because, irrespective of their stated commitments to the LMC, each GMB country has divergent concerns that cannot be reconciled by cooperation alone. An all-too-obvious example is transboundary resource management. Over recent decades, a major driver of economic growth across the GMB has been the exploitation by each country of its "own" natural resources and ecosystem services.<sup>9</sup> This approach has helped produce the decades-long expansion of national output (i.e., GDP). It has also generated serious, sustained environmental degradation.<sup>10</sup>

.....

....

<sup>8</sup> LMC Plan of Action 2018, especially Section 1.

<sup>9</sup> Le and McPherson 2015; Schmittman, Corvino, and Katagiri 2017. Ecosystem services are the benefits humans derive from ecosystems (MEA 2005).

<sup>10</sup> Un and So 2009; Vietnam, 2011, p. 2; Vietnam 2016, pp. 50, 90; Lao PDR 2011, p. 7; OECD 2013; Menon and Warr 2013; WB 2017b, pp. 28, 32.

Thus far, no GMB country has significantly modified its pro-growth strategy to prevent or minimize these adverse effects.<sup>11</sup>

GMB countries could move themselves off this counterproductive trajectory by collectively reframing their approach to regional development.<sup>12</sup> This cannot happen as the LMC is currently organized and oriented. It is, nonetheless, a task that China, as the dominant LMC partner, could promote. Such a response would represent a consistent international extension of the country's announced (domestic) desire to create an "ecological civilization."<sup>13</sup>

The paper has the following format. Section 2 describes features of the GMB and the LMC and notes why cooperation (alone, or primarily) is not a viable means of long-term organization or administration. Section 3 focuses on three LMC themes—water governance, connectivity, and environmental sustainability—to illustrate the key challenges and opportunities in promoting transboundary resource management. Section 4 discusses how China could lead the effort to improve transboundary resource management throughout the GMB, and Section 5 concludes by examining the next steps to move the analysis and dialogue forward. Two annexes elaborate topics raised in the text—cooperation and public policy, and adverse cumulative policy outcomes.

11 There are many studies describing the adverse effects of damming the Mekong. None of them has deflected China, Laos, and Cambodia from continuing this activity.

13 Adopted by the Communist Party in 2007 and now a central feature of the 13th Five-Year Plan (CCCPC 2016, Part X; Wang, He, and Fan 2014; Xinhua 2017a; Standaert 2017; Xi Jinping 2017, and 2018, Section IX). The expression has a number of interpretations ranging from "environmentalism with Chinese characteristics" to "authoritarian environmentalism" to a program that reinforces "the symbiosis between economic development and environmental protection" (Wang-Kaeding 2018).

<sup>12</sup> The closest that any GMB country has come to formulating a regional development program has been Laos, with its plan to become the "battery of ASEAN" (Roberts and Sager 2016; Hoogzaad et al. 2017; Thomas 2018). The GMS corridor initiative (Banomyong 2008, 2013; ADB 2011, 2016;2019; Kobayashi et al. 2017) has specifically focused on infrastructure as a driver of regional development.

# **SECTION 2: BACKGROUND<sup>14</sup>**

#### The Greater Mekong Basin

The GMB, covering an area of 795,000 square kilometers—slightly less than the combined areas of California and New Mexico in the United States and slightly more than Qinghai Province in China—is one of the most biologically and geologically diverse regions on Earth. The Lancang-Mekong River, with an average annual discharge of 475 cubic kilometers, flows roughly 4,900 kilometers from the Qinghai-Tibet Plateau to the ocean through the Mekong Delta in Vietnam. The basin has been rapidly transformed, particularly over the past four decades as each country across the GMB has focused on boosting national income, output, exports, and employment by exploiting what each of their governments see as an abundant supply of "cheap" natural resources.<sup>15</sup> The resulting expansion of irrigation, hydro-power facilities, plantations, logging, wild-capture fishing, aquaculture, and the intensification of crop and livestock production have degraded the basin's natural resources, and urban growth and industrialization have compounded the damage. Much of the environmental transformation is irreversible.<sup>16</sup> With current policies, these trends will continue, and they will most likely be accentuated by climate change.

China has engaged with its GMB neighbors through several arrangements, the LMC being the most recent. Since 1992, the country has been a full member of the Greater Mekong Subregion Economic Cooperation Program (GMS), which was

Some changes may be technically reversible, e.g., draining wetlands, although the remedial costs may be prohibitive. For others, such as land subsidence and coastal erosion, the biophysical transformations are permanent since they cannot be feasibly repaired with existing (or even foreseeable) technologies and financial capacities.

<sup>14</sup> A reviewer suggested that this section describe the methodology being used and highlight the situation in the GMB before the LMC. The first is premature and the second a digression. Our task is to identify issues promoted by the LMC that improve transboundary resource management. The methodology and analysis will follow from the issues, not vice versa. On the second point, many scholars have examined development performance across the GMB. We build on their contributions to examine China's future options.

<sup>15</sup> Natural resources are only "cheap" from a private perspective. Their social costs, measured as the permanent loss incurred when the *stock* of natural wealth is converted into *flows* of income (GDP), are often exceedingly high (Tarp et al. 2007; Costanza et al. 1997, 2014).

"founded on a common vision, goals and strategic thrusts for regional cooperation."<sup>17</sup> And since 1996, China has been a "dialogue partner" of the Mekong River Commission (MRC), which was created a year earlier when the Lower Mekong Basin countries signed the Agreement on Cooperation for the Sustainable Development of the Mekong River Basin.<sup>18</sup>

The LMC has two advantages for China. First, it is a convenient means of interacting with GMB countries on matters that directly contribute to China's development and security. The mechanism is independent of the Asian Development Bank, the principal sponsor of the GMS, and Western donor agencies that support MRC, and is therefore free of "outside" influence.<sup>19</sup>

Second, the LMC is consistent with China's "going out" policy.<sup>20</sup> Initially launched to help sustain economic growth, "going out" now encompasses the vision of a Silk Road Economic Belt and a Maritime Silk Road initiative.<sup>21</sup> Designed to stimulate international trade and investment and raise China's global status by connecting scores of countries across the globe, both of them are now part of the Belt and Road Initiative (BRI).<sup>22</sup>

As the gateway to China's neighbors within the GMB and Southeast Asia, Yunnan Province is the geographic focal point of the LMC.<sup>23</sup> Critical to China's campaign to "develop the west,"<sup>24</sup> Yunnan is the origin for the BRI's China-Indochina Peninsula

- 18 Cooperation is common to both arrangements. The MRC was formalized on April 5, 1995, when the "Agreement on the Cooperation for the Sustainable Development of the Mekong River Basin" was signed by the heads of government in Thailand. Article I, "Areas of cooperation," states: "To cooperate in all fields of sustainable development, utilization, management and conservation of the water and related resources of the Mekong River Basin including, but not limited to irrigation, hydro-power, navigation, flood control, fisheries, timber floating, recreation and tourism, in a manner to optimize the multiple-use and mutual benefits of all riparians and to minimize the harmful effects that might result from natural occurrences and man-made activities."
- 19 Some scholars stress the strategic motivation for the increased engagement (Zhou Shixin 2018).
- 20 *Economist*, 2009; Nash 2012; Wang Hongying 2016.
- Aoyama 2016; Bruce-Lockhart 2017; *Economist*, 2017; Tweed 2018; Tan 2019.
- 22 Truong-Minh Vu and Mayer 2018.
- 23 Despite Yunnan being its geographic base, the LMC is administered from Beijing.
- 24 Jin Kai (2014), Li Hongmei (2013), CEN (2013, 2019), Leibold (2014), and Singh (2016). Develop-the-West was introduced in 1999 by the central government to stimulate growth in the 12 western provinces. I thank Professor Anthony Saich for discussions on this topic.

<sup>17</sup> ADB 2011, pp. 2–6; ADB 2016a; ADB 2018.

CHINA'S ROLE IN PROMOTING TRANSBOUNDARY RESOURCE MANAGEMENT IN THE GREATER MEKONG BASIN (GMB)

Economic Corridor, with high-speed rail, modern roads and bridges, industrial parks, hydropower dams, and pipelines.<sup>25</sup>

#### Lancang-Mekong Cooperation

The LMC Plan of Action describes the LMC's development goals as:

contributing to the economic and social development of sub-regional countries, enhancing the well-being of the people, narrowing the development gap within the region and building a Community of Shared Future Peace and Prosperity among the Lancang-Mekong Countries.<sup>26</sup>

To meet these goals, the LMC will:

take into account the development needs of the six LMC member countries and the regional integration process, reflect the framework established in the Sanya Declaration, highlighting the leaders' guidance, all-round cooperation and broad participation, and follow a government-guided, multiple-participation, and project-oriented model.<sup>27</sup>

The "development needs" are political and security interests, economic and sustainable development, and social and cultural cooperation. The first category includes four items, the second ten, and the third six.<sup>28</sup>

The Plan of Action will be implemented "based on the principles of consensus, equality, mutual consultation and coordination, voluntarism, common contribution and shared benefits, and respect for the United Nations Charter and international laws, as well as in accordance with domestic laws, rules, regulations, and procedures of each member country."<sup>29</sup>

<sup>25</sup> Because of its location, its biodiversity, and its large hydropower capacity (with seven hydro dams and plans for up to two dozen more), Yunnan also has a central role in the 13th Five-Year Plan's quest to expand renewable energy, promote green growth, and foster environmental sustainability.

<sup>26</sup> LMC Plan of Action 2018, Section I.

<sup>27</sup> Ibid., Section II.

<sup>28</sup> Ibid., Sections 4.1, 4.1, and 4.3.

<sup>29</sup> Ibid., Section II.

These principles will create a new form of regional cooperation:

By synergizing China's Belt and Road Initiative and the ASEAN [Association of Southeast Asian Nations] Community Vision 2025 as well as the Master Plan on ASEAN Connectivity 2025 and visions of other Mekong sub-regional cooperation mechanisms, the LMC is moving towards a new sub-regional cooperation mechanism with unique features driven by internal strength and inspired by South-South cooperation, which will support the ASEAN Community building and regional integration process, as well as promote the implementation of the UN 2030 Agenda for Sustainable Development.<sup>30</sup>

Finally, the Plan of Action provided a time frame:

Years 2018 and 2019 will be the foundation-laying stage when emphasis should be focused on strengthening sectorial cooperation planning and implementing small and medium-sized cooperation projects. The years 2020–2022 will be the consolidation and expansion stage, when member countries will further strengthen the cooperation in the five priority areas and may explore new cooperation areas that help respond to the development needs of member countries, optimize cooperation model[s] and gradually explore cooperation on large projects.<sup>31</sup>

Several points stand out. The approach is all-encompassing, with references to the BRI, the United Nations' Sustainable Development Goals (SDGs), the ASEAN master plan, and other regional arrangements. The agenda is ambitious, comprising 20 topics related to politics, security, economics, sustainable development, and social and cultural cooperation. There is no ambiguity about its administration. The LMC's activities will be "government-guided, multiple-participation, and project-oriented," with their "implementation" based on the 10 "principles" quoted above.

These features would be heavy lifting for any arrangement, let alone one that does not define its principal term. "Cooperation" appears 98 times in the Plan's 14

<sup>30</sup> Ibid., Section I.

pages but is never defined.<sup>32</sup> Even if we assume the most common meaning of "working together," the Plan of Action leaves unanswered the question of whose cooperation is required, how, with whom, when, where, and over what period to reflect the "leaders' guidance" in following "a government-guided, multiple-participation, and project-oriented model"? Even if the term were defined, "cooperation" is an inadequate means for ensuring LMC activities are formulated and implemented. It offers no guidance on how differences in the participants' perceptions and preferences will be handled, by whom, based upon what criteria or, if they are to be reconciled, how any concessions will be granted by whom and to whom.<sup>33</sup>

The Plan of Action's authors left some wiggle room by noting the need to "further explore the most suitable cooperation model for the Sub-regional cooperation in line with the unique features and specificity of the six countries."<sup>34</sup> But suggesting that everyone will cooperate to devise means of cooperating simply pushes the problem back one level.

There are other concerns, especially as they relate to the implementation of the LMC agenda. The Plan of Action does not indicate how the proposed "new sub-regional cooperation mechanism with unique features" might operate. Outsiders—and even many insiders—lack relevant information on LMC projects, both ongoing and proposed.<sup>35</sup> The fragmentary data available show that projects range from training and

- 34 LMC Plan of Action 2018, Section II.
- 35 As an illustration, the LMC website www.LMCChina.org/eng provides only the most general information and press reports, all of which reaffirm the "warm" bilateral relations and positive outcomes of high-level engagement. Descriptions of what is being done by whom, at whose expense, and for whose benefit are not provided.

<sup>32</sup> This comment applies even to the Plan's sections on "working structure" and "practical cooperation" (Sections III and IV), both of which should have furnished some minimal idea of what the concept meant. Terms may be undefined because it was too difficult to reach a mutually acceptable meaning. Alternatively, the intention is strategic, enabling participants to use the meaning that best suits their purposes.

<sup>33</sup> The descriptions include "optimize the multi-layer framework," "strengthen communication and coordination," "promote dialogues and exchanges," and "enhance cooperation" without indicating how any of them can be done, by whom, or to what effect, or noting how differences of approach, if any, will be resolved. One reviewer suggested that the vague language may not create problems if LMC participants engage in "step-by-step" learning to determine what form of collaborative action might be possible. This is true, but it will be time-consuming and stretch the LMC foundation period (noted in the timeline) well beyond anything that its organizers anticipate. Moreover, to be effective, the learning from this engagement will need to be widely shared, something that the LMC does not encourage.

exchange visits that would cost thousands of dollars to high-speed railways and hydropower dams that would cost billions. When the LMC Secretariat was launched in November 2017, it was announced that 45 "early-harvest" projects were underway.<sup>36</sup> During 2018, other sources reported that 20-plus ongoing infrastructure and industrialization projects were being implemented and "more than 100" projects related to "quality of life" had been completed.<sup>37</sup> Another source stated that by early 2018, there were financial commitments for 132 LMC projects.<sup>38</sup> No details of these activities have been published, although some external efforts to derive them are underway.<sup>39</sup>

#### The Administrative Inadequacy of Cooperation

As described in Annex I, cooperation cannot and will not sustain the LMC over the long term. The problematic histories of GMB countries stretch back decades and even centuries, and cannot be diminished by extravagant, repeated commitments to cooperate. All GMB countries, including China, will cooperate only in ways that are consistent with their interests, goals, capacities, and preferences. To expect otherwise is naïve and a misreading of the region's history.<sup>40</sup>

There are too many facts on the ground that influence, even if they do not deter, mutual cooperation.<sup>41</sup> China's cascade of seven dams (so far) on the Lancang River

<sup>36</sup> Zhang Yue, 2017.

<sup>37</sup> Liu 2018; Lifang 2018.

<sup>38</sup> Nguyen Khac Giang 2018.

<sup>39</sup> These involve triangulating multiple sources to produce detailed estimates of China's development assistance and its impacts (Strange 2018). They show that from 2000 to 2014, China spent \$354.3 billion on "foreign aid." Roughly 20 percent of the expenditure meets the international definition of official development assistance (Junyi 2016; Murphy 2017; Hatton 2017). Problems arise regarding the development impact—on "economic growth," "debt sustainability," and "environmental sustainability"—of the remaining 80 percent. Ferchen 2018 reviewed these data and concluded that: "the finding that the majority of China's official finance is based on commercial terms and that it does not contribute to economic growth in the host countries should be as headline-grabbing as anything." Gaining access to details of what is unfolding in the GMB is likely to require a similar external effort.

<sup>40</sup> One reviewer asked "why should history matter here?" The short and long answer is "mutual mistrust" to which we return below.

<sup>41</sup> A reviewer argued that all of these were underway before the LMC. She inquired: What could be done under the LMC to deal with these issues? A brief answer is "nothing much, if anything," since the LMC's emphasis on "cooperation" excludes consideration of factors—such as interests, incentives, policies, and power—that drive the underlying behavior.

was constructed with no consultation regarding their downstream effects.<sup>42</sup> There are plans for up to 20 more.<sup>43</sup> The largest cities on the Mekong–Vientiane, Phnom Penh, Can Tho-dump untreated effluent into the river. Plans to treat urban effluents are limited and, at current rates, will not be operational for many years. Furthermore, since the relevant national agencies do not enforce existing environmental regulations, this pollution will most likely worsen. The lack of drainage in major irrigation systems in Thailand, Cambodia, Laos, and Vietnam worsens regional water pollution. None of the water authorities in these countries has plans to re-engineer their irrigation schemes to rectify this deficiency.<sup>44</sup> Logging and plantation development silt up waterways by accelerating hillside erosion. Uncontrolled logging remains a prominent feature of forestry "development" in Lao PDR and Cambodia.45 Overfishing continues to deplete wild-capture fisheries in the GMB's rivers and lakes. Regulations reducing overfishing are weakly enforced, in part because GMB governments have systematically underfunded their resource-protection agencies.<sup>46</sup> Scores more examples, many given in this paper, are available to illustrate the point that moving beyond the current circumstances through cooperation alone, or primarily, is fanciful. Methods that appropriately incorporate the interests and concerns of all GMB countries are needed.

Yet, even if the GMB partners were inclined to override their interests and preferences and cooperate to address the problems created by these (and other) facts on

42 Truong-Minh Vu and Mayer 2018.

44 On the few occasions when drainage is mentioned in the GMB literature, ti refers to urban areas and flood control (MRC 2004, 2012).

<sup>43</sup> IR 2013; Cronin 2014; Rasanen 2017; Bernstein 2017; Sullivan 2018. The primary function of the dams is energy production. They are managed to impound the maximum amount of water during the wet season to sustain electricity generation in the dry season. China derives economic benefit from the dams, but they have adverse downstream effects. They reduce the translocation of silt, which is important for lowland fertility, and fluctuations in the river flow affect the productivity of wetlands and related ecosystems. Several scholars have argued that the dams strengthen China's capacity for "hydro-diplomacy" (Middleton and Allouche 2016; Yeophantong 2016; Brilingaite 2017; Zhang and Li 2018), but this presumed advantage is exaggerated. The Lancang River watershed yields 18 percent of the Mekong's annual average flow and existing dams can impound 28 percent of that flow (about 23 cubic kilometers). Thus, water cannot be carried over from one hydrological cycle to the next. This point will remain relevant even when (or if) all planned dams are completed. China's threats to hold water back for diplomatic advantage are not credible.

<sup>45</sup> Terra Daily 2013; Butler 2013; Saunders 2014; IUCN/WCS/WWF 2017.

<sup>46</sup> Allan et al. 2005; Berdik 2014; Connor 2019; Eyler and Weatherby 2019; Seiff 2019.

the ground, their prospects for joint action would be blocked by a more formidable barrier, namely, the requirement that the LMC be based on "a government-guided, multiple-participation, and project-oriented model." This approach, which has been fundamental to China's worldwide "development assistance" for decades,<sup>47</sup> is unlikely to be modified to suit GMB partners. It is a top-down mechanism that treats development as something that China does *for* recipient countries and their people rather than something it does *with* them.<sup>48</sup>

This operational model has several drawbacks. First, it is explicitly noncooperative, characterized by a lack of local consultation on project selection and minimal levels of local participation in project implementation. Second, project funding is almost exclusively determined by what Chinese entities will finance rather than what the recipient country has the capacity to support.<sup>49</sup> Third, "government-guided" projects have a poor record of protecting the environment and boosting community livelihoods.<sup>50</sup> And finally, it does not promote regional development. This is obvious with the LMC, where China manages its development assistance through bilateral, high-level, government-to-government interactions. We return to this point below.

------

- 47 China has six categories of assistance: "undertaking complete projects, providing goods and materials, conducting technical cooperation and human resources development cooperation, dispatching medical teams and volunteers, offering emergency humanitarian aid, and reducing or exempting the debts of the recipient countries." (Xinhua 2011; State Council 2014, Section III; Carter 2017, Section 2). The first two items account for more than 80 percent of China's foreign assistance (State Council 2014; Yun Sun 2014; Junyi Zhang 2016).
- 48 Gaining access to information on China's foreign aid or official finance is difficult. Formally, it is a state secret. Yet, even if it were not, aid, official finance, and foreign investment are regularly conflated (Kitano 2016). Copper 2016 (p. xi) explained: "China has linked its economy to the rest of the world, providing China the key to expanding its status as a world power while forcing many to accept a redefinition of the terms 'foreign aid' and 'foreign investments.'" A recurrent issue is whether China's foreign aid has strings attached. Officially, there are no conditions (State Council 2011, Preface). Few observers believe this(Kitano 2016, section 2; Chong 2017; Fang 2018). Four of the top five recipients of China's official finance—Angola, Pakistan, Russia, and Venezuela are under China's direct influence.
- 49 Kynge 2018. The "debt trap" issue is discussed further below.
- 50 Mekong Eye 2016; Nguyen Khac Giang 2018; Wang Yan 2018.

## **SECTION 3: LMC THEMES**

This section relates transboundary resource management to the three LMC themes connectivity, water governance, and environmental sustainability. There are four reasons for emphasizing these topics. First, they are central to the LMC Plan of Action and, as such, have been formally agreed by all GMB countries. Second, they feature prominently in the national agendas of each GMB country. Third, none of them is handled well or consistently by GMB governments. Fourth, they all generate significant positive and negative transboundary externalities.

#### Connectivity

The value of upgrading and extending infrastructure for all GMB countries is beyond question;<sup>51</sup> there is compelling evidence that improving connectivity stimulates and sustains rapid economic and social development.<sup>52</sup> China's advance since the late 1970s is an example.<sup>53</sup> Based on that experience, China's leaders have concluded that large-scale infrastructure investments will be equally transformative on a global level. The BRI backs this view with action.<sup>54</sup>

Many of the infrastructure projects already underway as part of the LMC either began as, or have become, BRI activities. With few exceptions, they are being implemented by Chinese state-owned enterprises, with Chinese technical support, using commodities and materials sourced from China, and they are financed by official or semi-official Chinese entities. As mentioned earlier, this reflects China's long-established approach

<sup>51</sup> ADB 2005, 2012a; WB 1994, 2009.

<sup>52</sup> The remains of Roman bridges, roads, viaducts, aqueducts, and ports highlight this point. Scholars such as Deane (1965) have argued that the United Kingdom's canal system helped stimulate the Industrial Revolution.

<sup>53</sup> Prior to China's rise, the gains registered by the "Asian tigers"—Singapore, Hong Kong, South Korea, and Taiwan—depended heavily on broad-based improvements in infrastructure (WB 1993; Page 1994). Finally, all "high growth performance" economies identified in the World Bank's Commission on Growth had invested heavily in infrastructure (WB 2008, Brady and Spence 2010).

<sup>54</sup> Zhu Feng 2013; *Economist* 2017; Millward 2018. When asked about the inspiration for the BRI, President Xi stated that it was "aimed at galvanizing global cooperation and is open to all" (Xinhua 2017b). Other assessments of the BRI are more guarded (Kuo and Kommenda 2018; Small 2018; Kynge 2018; Bloomberg 2018; Greer 2018).

to development assistance.<sup>55</sup> Its major advantage is that it gets things done, even if after-the-fact monitoring reveals significant slippage.<sup>56</sup> But it often leaves recipient countries with several disadvantages related to project financing, the quality of work, environmental damage, local participation,<sup>57</sup> and sustainability.<sup>58</sup>

Infrastructure and connectivity across the GMB have improved over recent decades. Progress in some areas—urban roads,<sup>59</sup> electrification, air travel, and telecommunication coverage—has been dramatic, although much remains to be done. Multiple studies describe the massive investments required to upgrade GMB connectivity and highlight several areas needing special attention.<sup>60</sup> There is no regionwide electricity grid.<sup>61</sup> The road network has limited capacity and is undermaintained. There are few international or internal bridges. Ferry and barge services have low capacities, formal border-crossing points are congested, and transit procedures are costly and time-consuming. River port facilities cater to local traffic rather than long-distance trade and commerce. There is an east-west economic corridor,<sup>62</sup> but nothing approximating a north-south one.

Ongoing and proposed LMC connectivity investments will address some of these deficiencies, and LMC communiques suggest that billions of dollars have already been committed. This is a useful start, but other issues need to be addressed. GMB governments should consider whether the projects chosen contribute to a coherent regional development program, how the recurrent costs (i.e., operations and maintenance) will be covered once the investments are made, and if the investment projects contribute to narrowing the GMB's development gap.

- 57 Chandran 2018.
- 58 A recurring problem in Cambodia has been poor quality road construction (Serey 2012; Dara and Masy 2018).
- 59 Banomyong 2008, 2013; Perdiguero 2016; ADB 2011, 2016a, 2018.
- 60 Asia and the GMB's infrastructure gap have been widely analyzed. Estimates for Asia of the *capital* costs are around \$1.7 trillion a year (ADB 2017; Sungsup Ra and Zhigang Li 2018; WB 2018).
- 61 Despite the widely discussed effort to develop a GMB power pool (Antikainen, Gebert, and Moller 2011; ADB 2012b; Weatherby and Eyler 2016), a detailed assessment of the situation noted that there is "very little connectivity in the power systems of the GMS" (ADB 2013a, p. 86).

62 ADB 2010.

<sup>.....</sup> 

<sup>55</sup> French 2010. Referring to Chinese support to Africa, Yun Sun (2014) noted: "Chinese development finance, combined with the aid, aims at not only benefiting the local recipient countries, but also China itself. For example, China's 'tied aid' for infrastructure usually favors Chinese companies (especially state-owned companies), while its loans are in many cases backed by African natural resources."

<sup>56</sup> Facilities are built, but the quality is often substandard (*Economist*, 2011, 2015; Wang Ruka 2017; Greer 2018).

With respect to the role of infrastructure in fostering regional development, GMB countries have collaborated extensively in designing and building infrastructure throughout the region.<sup>63</sup> The LMC Plan of Action referenced this activity when it highlighted the ASEAN Community Vision and the Master Plan on ASEAN Connectivity. As noted above, the BRI would "synergize" with these programs. Thus far, nothing has been published to describe how that would be done, by whom, over what period or what effort, if any, the LMC would make to link with the ASEAN and GMS initiatives. With China deciding LMC activities on a bilateral basis, regional activities will not be supported. Thus far, LMC projects have been chosen to more closely match the capacities and preferences of the implementing agencies (primarily Chinese SOEs) rather than the projects that fit existing regional development programs.<sup>64</sup> High-speed rail facilities, coal-fired power plants, and industrial parks are obvious examples. For its part, China is motivated to improve the access of its western provinces to international seaports. Whether and how, if at all, this priority fits within the broader infrastructure expansion plans of GMB governments is yet to be determined.<sup>65</sup>

If GMB governments are to operate efficiently and maintain appropriately the anticipated expansion in LMC infrastructure investments, they will need to fully fund the additional recurrent costs.<sup>66</sup> Infrastructure investment adds to a country's physical capital stock, but the economy will continue to benefit from that increased capital

A review of the current national development plans for each GMB country shows that they cover issues such as international outreach, regional integration, and regional infrastructure, and in the case of China also the "deepening" of "pragmatic cooperation with neighboring countries" (CCCPC 2016, chapters 52 and 53). Other GMB countries note their commitments to the GMS and ASEAN regional development activities, particularly transportation (Lao PDR 2016, Section 3, especially 3.4.6, and Section 5, especially 5.3; Vietnam 2016, section 3, point 12 and Part V; NESDB 2017, p. 22, Part 4, Strategy 10, and Section 2.7.2; Myanmar, Government of 2018, p. 4, section 3.4.8 and 3.6.6; Cambodia, Royal Government of 2018 pp. 4, 5, and 17 and Section 1.3, Rectangle 2). Cambodia's plan is the only one that mentions "One Belt-One Road," or the BRI as it is now called (*op. cit.*, Part 1.6).

64 Tan (2019, p. 15) refers to an American Enterprise Institute study stating that "through the end of 2017, Chinese state-owned enterprises (SOEs) were responsible for 95% of BRI construction."

<sup>65</sup> As of December 2019, the LMC website offers no guidance on this point. Its "top news" and "bilateral cooperations [*sic*]" refer to a host of individual activities, none of which involve multilateral or regional planning.

<sup>66</sup> Some infrastructure, such as Cambodia's roads, mentioned above, has both poor initial construction and limited maintenance. Public infrastructure in Vietnam has suffered from a chronic shortage of recurrent-cost financing for years. A public finance review noted: "the] maintenance budget can only meet approximately 50 percent of estimated needs at the moment" (WB/VNM, 2017, pp. 15, 19).

only so long as it is operated at, or near, its installed capacity.<sup>67</sup> Two matters require attention. First, each government should maintain an appropriate balance between its *capital* spending and the annual flow of *recurrent* resources needed to operate and maintain the physical facilities. It will also have to generate the human capabilities to manage and operate the capital projects—hydro dams, telecommunication facilities, roads, bridges, power plants, pipelines, railways, effluent treatment facilities, and industrial zones. Under normal circumstances, these capabilities are created when relevant professionals and workers are trained while the infrastructure is being constructed. That, however, requires participation, something the LMC has not promoted.

LMC activities will encounter recurrent-cost problems. None of the GMB governments regularly fund their recurrent costs adequately.<sup>68</sup> Indeed, they typically support infrastructure investment because it offers tangible evidence of economic progress, but they are less enthusiastic about having to make the annual budgetary allocations to operate and maintain the additional capital improvements. Consequently, the capital regularly performs below its designated capacity,<sup>69</sup> and those who use it incur higher costs resulting from added wear-and-tear, congestion effects, and reduced reliability.<sup>70</sup>

- 67 Utilization rates are critical determinants of an investment's viability. If operational inefficiencies or poor maintenance prevent planned utilization rates from being attained, prospective benefits are lower than projected and user costs are higher. Examples of Chinese investment, old and new, illustrate this. The Tazara railway, the first major Chinese investment in Africa (built during the early 1970s), has operated well below capacity for decades (Schneider 2010; van Mead 2018). Kenya's Standard Broad Gauge Railway was significantly overbuilt (*Economist* 2015, Oirere 2016, Kacungira 2017), and since it was completed in 2017 has operated at less than half its projected capacity (Griffiths 2018).
- 68 There is nothing novel about the recurrent-cost problem. Providing adequate recurrent funding is a fundamental principle of responsible public finance. Failing to provide for maintenance was a major cause of wasted capital in the Soviet Union (Granick 1957). The problem was also widely studied by aid agencies in the 1970s and 1980s, especially in Africa. Donors would invest in roads, bridges, buildings, irrigation systems, and so on, but local governments did not provide the resources to operate and maintain them (Heller 1979; Gray and Martens 1982). That pattern has persisted.
- 69 Performance is typically measured as vehicles per hour for roads, bridges, and tunnels, rush-hour passengers for ferries/buses/trains, and peak-demand volume for pipelines.
- 70 There are many examples of unrepaired or poorly maintained roads increasing vehicle operating costs and travel times; intermittent electricity supply raising investment costs (e.g., in back-up generators) and reducing the serviceable life of electrical equipment; and undermaintained and poorly supplied public facilities such as schools, hospitals, and health clinics lowering the quality of learning, raising public health risks, and reducing citizen access to public services. One source argues that the decline in China's investment in infrastructure will reduce economic growth, but this is only partly true (*Economist 2018*). Growth will also decline unless existing capital is appropriately maintained and operated.

The third matter related to connectivity is the degree to which the relevant investments contribute to the LMC's development goals. Properly selected, planned and implemented, connectivity investments would help narrow the development gap, promote economic and social development, enhance citizen welfare, and build a regional community of shared peace and prosperity. Prudent management is needed to achieve these outcomes. Excessive levels of debt to finance the infrastructure will widen rather than narrow the development gap with a debt trap, blocking or reducing further investment.<sup>71</sup> Citizen welfare will not be enhanced if infrastructure investments do not meet pressing community needs. As an illustration, when completed, the China-Laos railway will provide ultra-modern high-speed transport for those traveling from Kunming to Vientiane and beyond.<sup>72</sup> This billion-dollar-plus project will speed passengers past towns and villages without electricity and other basic services, delivering them to a city that lacks a functioning wastewater treatment facility. Such contradictions should be avoided if citizen welfare is to improve and the development gap to narrow.

Promoting economic and social development requires attention to local engagement in decisions about what infrastructure will be constructed and local participation in its construction and maintenance. This is where the LMC's "government-to-government" decision-making breaks down. "Trickle-down" effects are weak, at best.<sup>73</sup> LMC project selection should ensure that significant numbers of GMB citizens and households contribute to, participate in, and benefit from LMC activities. A regional community of

.....

72 Xinhua 2019.

China's foreign minister was quoted as denying that China's assistance contributes to a "debt trap" (Chen Jia 2019). Zeinullayev 2019 (Section 4.1, p. 15) specifically found that "no [BRI] country has crossed the high debt threshold purely due to BRI lending." Both ignore a fundamental lesson about debt, namely, avoiding debt problems requires responsible borrowers *and* responsible creditors. Zeinullayev's data reveal that BRI lending has significantly augmented the debt of countries (Djibouti, Kyrgyzstan, Egypt, Ukraine, Sri Lanka, Pakistan, Montenegro, Iraq, Belarus, Laos, Maldives, and Mongolia) that had debt above the "sustainability" criterion of 60 percent of GDP.

<sup>73</sup> The Western development strategy mentioned earlier is an example of "trickle-down economics." It has had a modest but unbalanced impact with small declines in the relative income/welfare gap but large increases in the absolute income gap (Moxley 2010; van Dijk 2011; Kwong Man-ki 2015). The China Statistical Yearbook 2018, Table 6–17 shows that from 2013 to 2017, average household disposable income in Tibet (the poorest Western province) increased from 23.1 to 27.1 percent of average household disposable income in Shanghai (the richest in the East). By contrast, the absolute income gap between the two increased from ¥32,433 in 2013 to ¥42,977 in 2017.

shared peace and prosperity will emerge if the benefits of connectivity are broadly disseminated. But for this to materialize governments should actively strive to minimize the negative effects of infrastructure expansion, particularly as it relates to population displacement without compensation, environmental damage, and the continued neglect of welfare programs of groups whose villages and towns remain unconnected. Finally, shared peace will be elusive if the social costs of the infrastructure-generated prosperity are too highly concentrated, especially on the poor.<sup>74</sup>

#### Water Governance

Water governance relates to the efficient administration and management of all water resources—blue, green, and gray, surface and subsurface.<sup>75</sup> No GMB country administers water efficiently. This is puzzling since the wealth and prosperity of all six countries depend directly on the private benefits, social amenities, ecosystem services, and economic productivity that water provides.

Ensuring that water makes the maximum possible contribution to national and regional development would involve rationalizing the current water-management arrangements, creating basinwide procedures to price water more appropriately, and raising the quality of water throughout the GMB. A fourth goal would be to improve the access of the basin's poorest households to safe water.<sup>76</sup>

Water is not managed efficiently across the GMB because the formal authority for water supply, control, quality, access, distribution, marketing, drainage, and pollution abatement is fragmented within and across dozens of ministries and departments.<sup>77</sup>

<sup>74</sup> Apart from specialists, few officials notice the effects of dam construction and operations on daily and seasonal water flows. The disruptions and variations in flow reduce the productivity of wetlands and damage river-bank plots (Clark 2014; Laura Zhou 2018). The poor—most often the very poor—derive a large share of their liveli-hoods from wetlands and riverbanks.

<sup>75</sup> It is also defined as "what determines who get water when and how much" (UNDP/SIWI 2016, p.4).

<sup>76</sup> This is part of Sustainable Development Goal no. 6, to which all GMB governments are committed.

<sup>77</sup> Country reports compiled by Aquastat, the FAO's global information system on water and agriculture (available at www.fao.org/nr/water/aquastat/countries\_regions/) have sections on institutions and water management. In 2012, Laos had 10 separate committees and departments under four ministries and the prime minister's office that were responsible for "water management, policies and legislation related to water use in agriculture." For Vietnam, the 2011 report describes water-related entities within nine ministries and the PM's office plus the National Water Resources Council. The 2011 report on Thailand noted "in total there are 31 ministerial departments under 10 ministries, one independent agency and six national committees that are involved in

The problem is worsened by the lack the incentives to collaborate. With authority so widely disbursed, relevant laws and regulations are (at best) weakly enforced.

The national-level divisions are repeated and often magnified at the subnational level. For countries like China and Vietnam, with administrative structures stovepiped from the central to the subcommune and village levels, this web of overlapping authority leaves key tasks untended. Maintaining water quality is an example. There is no incentive for China and Laos, both of which are upstream, to change their behavior to preserve downstream water quality. This problem recurs at the national level as well. Water users in the Upper Mekong Delta of Vietnam pay scant attention to how their behavior—the over-application of agrochemicals, high levels of water extraction, and aggressive flood control—multiply the difficulties faced by the population in the lower delta.

Despite the myriad national agencies involved in managing water, no GMB country has an agency or group of collaborating agencies responsible for transboundary water governance. The Mekong River Commission has no mandate in this area either. As stated in the 1995 agreement cited earlier, its role is to foster the development of "an economically prosperous, socially just and environmentally sound river basin," which it has been pursuing through detailed surveys of the basin's features, analyses of water-supported livelihoods, studies of factors affecting river flow (such as damming and climate change), the selective monitoring of water quality and seasonal flood

..... water resources development." In 2011, Cambodia had four ministries and four separate units and departments handling water issues. The 2014 report for Myanmar stated that "no institution is responsible for the overall management of national water resources in the public and private sector." The Ministry of Agriculture and Irrigation, the Water Resources Utilization Department, the Irrigation Department, the Settlement and Land Records Department, and the Agricultural Planning Department all have some jurisdiction over water. Finally, a 2012 Aquastat report for China showed that seven ministries, two national commissions, the state council, and local water resource management departments and water resource bureaus oversee various, often related, aspects of water. A Global Water Partnership report (GWP 2015) highlighted the urgency for China to adopt an integrated approach to water management. A more recent report indicated that there are now 11 ministries involved in water management and protection (Zhu and Kong 2016). Some GMB countries are attempting to overcome this fragmentation. In Vietnam, the Ministry of Planning and Industry intends coordinating all water activities, although the details are still being worked out (interview with MONRE officials Hanoi, April 2019). Thailand, too, plans place the 38 governmental agencies concerned with water under an Office of National Water Resources (ONWR). The methods for doing this are being devised. Myanmar has recently formed a National Water Resources Committee. (I am grateful to Chayanis Krittasudtacheewa for this information.)

pulses, and training scores of officials and others on water-related issues.<sup>78</sup> None of the MRC's activities has addressed the issue of how water in the Lower Mekong Basin is, or should be, managed and administered, and by whom, however.<sup>79</sup> Other international organizations, such as the International Water Management Institute (IWMI) face similar issues.<sup>80</sup> and there is no indication that this situation will change.<sup>81</sup>

There have been consultations over water issues between the MRC and LMC. They have agreed on six areas for cooperation, and the MRC has suggested transboundary environmental assessment guidelines, but concrete actions have been slow to emerge.<sup>82</sup> This gap leaves problems such as upstream/downstream implications of dam construction, overfishing, improving water quality, preventing water pollution, regulating sand-mining, and preserving or restoring wetland ecosystems widely recognized, well understood, but largely unaddressed.<sup>83</sup> Due to this institutional gap, there no obvious focal point for basinwide responses to the intensifying effects of climate change,<sup>84</sup> two of which are projected to be increased flooding and extended periods

<sup>78</sup> MRC 2004, 2005, 2009, 2012, 2015; Bache et al. 2012.

<sup>79</sup> The recently created Lancang-Mekong Water Resources Cooperation Center reflects this. It will deal with information sharing, flood control, drought releases, and the monitoring of hydrological data (Zhong Yong 2018).

<sup>80</sup> International water conventions do not help. Vietnam is the only GMB country that has acceded to the United Nations Watercourses Convention (UNWC).

<sup>81</sup> Even under the best of circumstances, China's GMB neighbors will have difficulty inducing cooperation in the area of water governance. China is the world's most "upstream" country, sharing 110 rivers and lakes with 18 downstream countries (Daming He et al. 2014; Zhang and Li 2018; CWR 2018). Because of this, China has no obvious incentive to agree with its GMB neighbors on water governance principles that could potentially bind it in ways that may be inconvenient in its dealings with other downstream countries.

<sup>82</sup> Again, I thank Chayanis Krittasudtacheewa for information on these areas of cooperation and for references to MRC's plans on environmental impact assessment, joint action plan, and joint monitoring plan for water resource cooperation.

On these matters, GMB countries could learn from international experience. There are roughly 270 transboundary river basins worldwide and much cumulative experience on transboundary water governance (Solanes and Gozalez-Villarreal 1999; Abbott et al. 2007; Zeitoun, Goulden, and Tickner 2013; Baten and Titumir 2016; OECD 2015, 2018). For this learning to occur, the LMC would need to support open, transparent research on the relevant water governance issues.

A productive area of collaborative research for all GMB countries would be to compile a water balance for the basin that tracks annual inflow and outflow. This would show the aggregate sources of water (surface and subsurface, blue, gray, and green) and their uses (irrigation, aquaculture, wild-capture fishing, hydropower, transport, industrial and urban use, household consumption), and functions (environmental health, preservation of wetlands, and provision of other ecosystem services). Once assembled, this information would enable researchers to trace the projected seasonal shifts in the water balance as climate change intensifies (MRC

of low dry-season river flow. Both will significantly affect the progress LMC makes toward its development goals.

Promoting the efficient use of water has several direct advantages. It raises the *effective* supply of water for key functions such as supporting ecosystem services. It broadens household access to water, thereby improving equity. It raises water productivity and creates an incentive to maintain water quality. Generating and sustaining these effects will require water users to change their behavior. Raising the price of water to levels approaching its social cost will be essential.<sup>85</sup> Making this distinction between price and cost is critical. The price of water is the exchange value mutually agreed by willing buyers and sellers,<sup>86</sup> whereas its social cost is the value of all resources (human, physical, and natural) employed to make it available to its users.<sup>87</sup>

Following the suspension of irrigation fees in Vietnam in 2008, the formal price farmers pay for irrigation water has been zero. The social cost of water, however, includes the cost in material, time, and effort farmers employ to gain access to the water at its delivery point, plus the value of all the resources they devote to using the water and disposing of any drainage or "waste."<sup>88</sup> These costs are generally low so farmers have no incentive to economize. Farmers gain but the aggregate (nationwide) impact is negative. An adverse aggregate effect is low water productivity.<sup>89</sup> Indeed, national

<sup>2009;</sup> ADB 2013b; Lebel et al. 2014; CCCPC 2016, chapter 21; Vietnam 2016, Section III; Schmittmann, Corvino and Katagiri 2017; Hong and Johnson 2018).

<sup>85</sup> A reviewer wrote, "Water price is a very sensitive issue in LMC countries. [It is] difficult to implement in practice." Both are true, reaffirming why "cooperation" is inadequate. Some entity will have to impose higher water fees and enforce them. This is a matter for public policy (Buchanan 2003).

<sup>86</sup> The number of buyers and sellers is less relevant than the voluntary nature of the transaction.

<sup>87</sup> The social opportunity cost is the value of all resources employed in their best alternative use (Buchanan 1998).

<sup>88</sup> A full accounting would include the amortization on the construction of the canal, the recurrent costs of maintaining and managing the water facilities, the foregone value of the water in its alternative uses such as ecosystem and transport services, aquifer recharge, biodiversity maintenance in wetlands, and household and industrial water supply. It would also allow for negative externalities, such as losses from waterborne diseases and pollution created by effluents and toxic substances.

<sup>89</sup> The GMB literature regularly confuses water use efficiency and water productivity (Dang Kieu Nhan et al. 2011; Dang Kieu Nhan 2013). Efficiency refers to the use of water to the point where its marginal value product (product price multiplied by the marginal physical product of water) equals its price. Productivity is physical output per unit of input, e.g., kilograms of rice per cubic meter of water. Vietnamese use water significantly less productively than, for example, their counterparts in China and India (WB 2016; WB/VNM 2017, p. 19; Bastiaanssen and Cai 2018).

output and income in Vietnam would rise if water were transferred to higher value uses such as industry and aquaculture. Social welfare would increase as well because the water "freed-up" would help sustain wetlands and coastal mangroves (reducing coastal erosion), flush iron sulfides from the soil, recharge shallow aquifers, and increase dry-season water flow, thereby reducing crop losses from saline intrusion.<sup>90</sup>

Since water charges are well below the social value of water in other GMB countries as well, major improvements in water allocation and water use would occur if they too raised their charges. Some observers have argued that access to water is a "right" and its price should be kept low to help the poor. In practice, the poor already have limited access to good-quality water and regularly bear significantly higher costs mainly waterborne diseases arising from consuming polluted water. The irony is that higher water charges, on average, help the very poor by increasing the water available for wetland preservation, supporting higher levels of wild-catch fish and aquatic production, and more frequently flushing the canals and pools from which the poor draw their household water.<sup>91</sup>

The general inability of GMB countries to maintain water quality represents a major governance failure. Households, towns and cities, industries, and mines regularly dump untreated effluents into the basin's rivers and lakes, and farmers, livestock producers, and aquaculture processors add agrochemicals and other contaminants.<sup>92</sup> The result has been a marked decline in overall water quality with several notable hotspots such as Vientiane, Phnom Penh, and the mid-to-lower Mekong Delta.<sup>93</sup> The increase in water pollution is tangible evidence that none of the GMB governments attaches a high priority to water quality, either within or beyond their jurisdictions.<sup>94</sup>

91 The very poor can have their access to water subsidized through a "lifeline" rate. Common to electricity distribution systems, a lifeline rate involves a low or zero charge for water consumption below a predetermined amount (e.g., 100 liters per person per day) with significantly higher charges for amounts beyond that limit.

<sup>.....</sup> 

<sup>90</sup> A reduction in upstream water use in agriculture improves the welfare of coastal residents. The increased river flow reduces their purchases of treated water and/or the costs of upriver travel to collect fresh water.

<sup>92</sup> A major study from the Government of Vietnam and World Bank highlighted the damage created by agricultural pollution in Vietnam (Nguyen Tin Hong 2017; Nguyen Van Cong 2017; Tung Xuan Dinh 2017).

Hart, Jones, and Pistone 2001; Kummu and Sarkkula 2008; Chea, Grenouillet and Lek 2016; Sopheap 2016;
 *Economist*, 2016; Meyn and Dara 2017; Chanthavilay et al. 2017

<sup>94</sup> The governments of Vietnam and China have both proclaimed that they seek clean water and unpolluted air (CCCPC 2016, Box 17; WB/MPI 2016, pp. 30, 106). President Xi Jinping has been widely quoted as saying, "Clear

China has recently introduced regulations to improve water quality.<sup>95</sup> Reports suggest some positive outcomes,<sup>96</sup> although decentralized implementation and regulatory indulgence are slowing progress.<sup>97</sup> The reality is that few (if any) of these circumstances would have arisen if GMB countries had enforced their existing anti-pollution laws and regulations.

Current institutional arrangements virtually guarantee that water quality will continue to decline. No GMB country includes the full social costs of water degradation in its project analyses or planning exercises.<sup>98</sup> This omission artificially inflates the financial and social returns for GDP-raising activities that degrade water quality. China's recent push to create an "ecological civilization," noted above, and efforts to "green the BRI" would help reverse that bias, if they were implemented.<sup>99</sup>

Other factors affect water quality—siltation from logging operations, urban and industrial dumping, groundwater contamination through overextraction, and increased salinization of coastal waters as dry-season water flows decline. Governments need to find remedies. Some immediate relief would result if GMB countries enforced their existing environmental laws that prohibit water-polluting activities. These actions

.....

waters and green mountains are as good as mountains of silver and gold." (Geall 2017; Yanzhong 2018). Yet neither country has created a viable path from present goals (of rapid GDP growth with continued pollution) to the clean water and clear blues skies that, in Vietnam's case, are expected to materialize by 2035. China's growing emphasis on renewable energy is a useful start. Yet, senior officials show little evidence that they recognize how long it will take for the "clear waters and green mountains" referred to by President Xi, to materialize. International experience—from the U.S., U.K., and Europe—indicates that it will take several decades. In Vietnam, key measures of pollution continue to worsen (WB 2017a; VNNet 2017; VOA 2018; DTiNews 2017).

- 95 In 2015, China introduced the Water Ten Plan, under which 12 ministries and departments were to be coordinated to undertake 238 specific actions in 10 major areas (CEN 2015).
- 96 In private communication, Chayanis Krittasudtacheewa noted that these outcomes were reported to the forum on environment of the China ASEAN Environmental Cooperation Center (CAEC) in 2017.
- 97 Zhang Jiaqi 2018.
- 98 A reviewer commented: "No country in the world does this." Besides being untrue, it misses the point. As von Ciriacy-Wantrup (1952, 1961) and the Sarkozy Commission (Stiglitz, Sen, and Fitoussi, 2009, p.7) pointed out we regularly ignore or devalue what is not measured. More important, as the commission noted, "what we measure affects what we do."
- 99 MEP 2017. Like the LMC Plan of Action, this initiative relies on cooperation. It includes a list of 25 projects ranging from "international high-level dialogue on ecological and environmental cooperation" to "China-ASEAN environmental technology and industrial cooperation demonstration bases." It does not explain how the cooperators will handle divergent objectives, perspectives, perceptions of risks, capacities to contribute, and willingness to participate.

should be complemented by an increased focus on agricultural drainage and the introduction of water fees. Fees would induce users to economize, thereby making water available for supporting ecosystem services that dilute and remove contaminants.<sup>100</sup> In some cases, households would need to be relocated away from saline-affected and heavily polluted areas.

#### Environmental Sustainability

It is difficult, both technically and conceptually, to define what environmental sustainability means across the GMB. Most of the Basin's population depends directly on the use of natural resources for their livelihoods.<sup>101</sup> They have dammed rivers, drained wetlands, expanded irrigation, migrated to urban areas, constructed infrastructure, extended logging operations and plantations, intensified aquaculture and livestock production, and discharged effluents into the environment. These actions have permanently transformed the immediate locations where they occur and often seriously affected areas downstream, downwind, or downslope.

Accordingly, any workable concept of "environmental sustainability" needs to be flexible.<sup>102</sup> To anchor the present discussion, we use the Google Dictionary definition define the environment as "the natural world, as a whole or in a particular geographical area, especially as affected by human activity." Sustainability, in turn, conveys the notion of persistence, endurance, renewal, regeneration, and revival.<sup>103</sup> Given projected increases in population and income, and anticipated rates and patterns of socio-economic development across the GMB, human-induced disturbances will continue and many will intensify. Government incentives, regulations, and investments will amplify or attenuate their impacts by shaping the behavior of

<sup>100</sup> MEA 2005.

<sup>101</sup> Accurate estimates of the GMB population are elusive. An Aquastat report (see note 77 above) cited 70 million in 2007. The MRC regularly reports that there are 60 million in the Lower Mekong Basin.

<sup>102 &</sup>quot;Sustainability" is a slippery term. Dobson (1996) reported more than 300 definitions in the literature, and major studies of sustainability provide detailed analytical frameworks to track its multiple dimensions (Matson, Clark, and Andersson 2016, chapters 1 and 2). Other scholars distinguish weak from strong sustainability (Hediger 2008; Davies 2013), with the distinction resting on whether human and physical capital can substitute for the loss of natural capital.

<sup>103</sup> Google Dictionary defines it as avoiding "the depletion of natural resources in order to maintain an ecological balance."

the basin's residents, particularly whether they continue to exploit or begin to preserve its natural resources.

Though the LMC Plan of Action does not define environmental sustainability, it affirms the commitment by GMB governments to the UN's Sustainable Development Goals, from which we can infer what environmental sustainability might mean for them. SDG 15 asserts that participating governments will "protect, restore and promote sustainable use of terrestrial ecosystems, sustainably manage forests, combat desertification, and halt and reverse land degradation and halt biodiversity loss."<sup>104</sup> Three specific actions by GMB governments would make a major contribution toward this goal. They should implement fully existing environmental regulations, appropriately value natural resources and ecosystem services, and manage collectively the basin's ecosystems as coherent entities.

The ineffectiveness of environmental protection across the GMB is perplexing. Existing legislation in each country stipulates how individuals, corporations, and other entities will avoid environmental harm and, if they create any, the procedures required for repairing and/or ameliorating the damage. Through strategies and plans and representations at multiple international gatherings, each GMB country has reaffirmed the critical importance it attaches to environmental protection.<sup>105</sup> But the problem is not passing legislation and devising relevant regulations; it is implementation. Oversight agencies are typically underfunded relative to their responsibilities, their staffs are inadequately trained and equipped, and they are regularly required to overlook the adverse environmental impacts of government and state-owned agencies. Environmental protection at best, is weakly, unevenly, and arbitrarily enforced. Thus, *de facto* policy (i.e., what is implemented) differs in major ways from *de jure* policy (i.e., what is intended). Through their actions, GMB governments have demonstrated that environmental sustainability has not been a priority. Moreover, even when there have been attempts to remedy the situation, implementation has been distorted by adverse incentive effects (see Annex II for an example).106

<sup>104</sup> United Nations 2015.

<sup>105</sup> For instance, see State Council (2015), CCCPC (2016, Part X) and Vietnam (2012a, 2014). Despite years of worsening pollution, China's 1989 environmental protection law was revised only in 2014 (Duggan 2014).

<sup>106</sup> Yanzhong Huang (2018) highlighted the contradictions involved in China's implementation of "pollution control." In the wake of Beijing's pollution emergency in January 2013, scholars criticized the folly of promoting

Natural resources across the GMB nations' ecosystem services are systematically undervalued. This practice has grossly distorted the allocation of all productive resources, leading to the destruction and degradation of large amounts of the GMB's natural wealth. The distortions have been perpetuated through the continued use of conventional valuation techniques that underrepresent the contribution of natural wealth to national income and welfare.<sup>107</sup> None of them includes the permanent loss of natural wealth required to generate each project's transitory flows of income.

To remedy this and other evaluation biases, each GMB country should begin basing its project and program decision-making on the full social contribution to national income and welfare of all productive factors, including natural resources and their associated ecosystems services. One method that is being increasingly adopted internationally is the System of Environmental and Economic Accounts (SEEA). These accounts, which upgrade the current System of National Accounts (SNA), explicitly value the stocks of natural, physical, and human capital and the flows of income they generate.<sup>108</sup>

Revamping national accounting systems takes time—particularly to generate the necessary human capacities and strengthen the relevant institutions—but some modest adjustments would have immediate desirable results. One is to introduce payments for ecosystem services (PES) on a broad range of natural assets. Vietnam, Thailand, and China are already doing this for selected services. Laos and Cambodia are moving in that direction. Examples include ecotourism charges, watershed service fees, irrigation fees and groundwater charges, an eco adjustment to utility bills, timber stumpage fees, and license fees for the regulated discharge into the environment

GDP, which is valued, at the expense of the environment, which is not (Wharton 2013; Albert and Xu 2016; Yana, Andersson and Shiqiu 2016).

<sup>107</sup> Tarp et al. 2007; ICEM/IUCN 2013; Lebel et al. 2014; Hall and Manorom 2015. Benefit-cost techniques, as well as the international System of National Accounts (SNA), attribute significant, but varying, contributions of unmeasured or "uncosted" inputs to the measured effects of physical and financial capital and human labor, artificially inflating their returns. This makes projects appear financially viable, irrespective of their impacts on natural capital.

<sup>108</sup> An advantage of shifting to SEEA is that it directly reveals the increasing relative scarcity and social value of natural resources. As Johnson (2000) noted, for all practical purposes, the stocks of natural resources and flows of ecosystems services are fixed; i.e., they cannot be augmented with current technology and financial capacities. These circumstances do not apply to human, physical, and financial capital or information. As the latter stocks cumulate over time, the value of natural capital (technically its shadow price) increases.

of effluents, such as treated water, exhaust from electricity-generation plants, and vehicle emissions.<sup>109</sup> The objective should be to charge for the services provided by natural resources—through fees, taxes, or direct pricing—at levels that approximate their social value/cost.

Another immediate constructive change would be for GMB governments to mandate that all project and program evaluations cover the full social costs of the natural resources being transformed (making allowances for the losses of amenity and third-party productivity) and projected remediation costs. Relevant procedures are widely available.<sup>110</sup>

None of these changes will alter the full social costs of investment activities, which are incurred whether governments acknowledge them or not. The changes will attribute the explicit contribution to output and income of natural resources, the value of which now enters conventional project appraisals at zero. This information will not radically alter the projects that are currently underway or have been selected for political reasons. But by focusing on the full social benefits of natural resources, government decision makers and other stakeholders (including citizens) will be reminded of the contribution natural resources and their associated services make to national wealth and welfare.

Transboundary effects can be readily incorporated into project and program appraisal as GMB countries upgrade their accounting procedures. These effects are characterized by the interdependencies, both positive and negative, for which the SEEA was designed.<sup>111</sup> Examples include the change in river flows and wetland productivity resulting from dam construction and operation, the distributional effects of expanding fossil-fuel electricity production, changes in agricultural productivity due to reductions in silt deposition, and the basinwide impact of the destruction of fish stocks through overfishing.

.....

<sup>109</sup> The revenue generated pays those who provide or sustain the services (Kolinjivadi and Sunderland 2012; Janekarnkij and Polpanich 2014; Lebel, Wattana, and Talerngsri 2015).

<sup>110</sup> Many of the methods were developed by the World Bank's efforts to measure the value of natural capital and develop a broader measure of national wealth (Dixon and Hamilton 1996). Much of that work has been incorporated in the Wealth Accounting and the Valuation of Ecosystem Services (WAVES) and other initiatives designed to improve natural resource valuation (WB 2012b, 2015; Vardon 2014).

<sup>111</sup> Van der Linde 2013. This is evident in Environmental Impact Assessments that Campbell et al. (2015) show focus on a narrow range of issues. The SEEA overcomes this bias.

The integrated management of the GMB's natural resources and ecosystems is a topic that links the work of many agencies, including the MRC, the GMS, and nongovernment organizations such as the International Union for Conservation of Nature, the World Wildlife Fund, SUMERNET, and the Asia Foundation. Most efforts to foster integrated ecosystem management have been thwarted by the GMB countries' views regarding sovereignty over their "own" natural resources. Whether these claims ever made sense, given the trans-frontier mobility of flora, fauna, surface and subsurface water, and the atmosphere, they are irrelevant as climate change intensifies.<sup>112</sup>

Constructive collaboration in this regard will require efforts to overcome the widespread institutional fragmentation that hinders environmental management.<sup>113</sup> Convincing GMB governments to give greater weight to the social value of natural resources will take time to make the case and to organize the response, for example by expanding PES and adopting SEEA. Creating the domestic capacities to deal with transboundary ecosystems will involve reordering institutional responsibilities and devising incentives consistent with those responsibilities. The bureaucratic difficulties, especially in authoritarian systems, should not be underestimated. Sharing information would provide a useful starting point.<sup>114</sup> An area where cooperation may be the easiest is the one furthest from central government such as community-run programs along national borders.<sup>115</sup>

The difficulties involved in each area will emerge as efforts to collaborate unfold. This is where the activities of international and regional agencies are important. These entities—some of which were referred to earlier—view natural-resource management

Raising the productivity of energy use (ADB 2013a) is an area where all GMB countries could make a significant contribution. A common global measure of energy productivity is output measured in 2011 constant international dollars per kilogram of oil equivalent (www.wdi/worldbank.org/table/3.8). Except for Myanmar, all GMB countries are below the world average of 7.9 in 2015. China was 5.7; Thailand and Cambodia were 7.5, and Vietnam was 7.7. (Data for Laos were not reported.)

<sup>113</sup> This is a key feature of Transboundary Natural Resource Management (TBNRM). Effective TBNRM is "dependent on good NRM; it is not meant to replace it" (van der Linde et al. 2001, p.106). So far, no GMB country has "good NRM."

<sup>114</sup> As Clark (2014) noted, it would avert downstream damage from unexpected water surges.

Based on lessons using TBNRM in transboundary peacebuilding, Roberts (2003) noted that starting locally builds trust, improves communication, sorts out land tenure issues, provides resources to meet local challenges, and enhances management capacities.

in transboundary terms. They have different perspectives, priorities, and time horizons, but these differences could be readily accommodated if GMB governments signaled their commitment to better manage their transboundary ecosystems. A final question is whether the GMB has an obvious leader in this area. Thailand has emphasized the "sufficiency economy" for more than two decades<sup>116</sup> and, as noted earlier, China has been promoting an "ecological civilization." In theory, either country could lead. That, however, is incidental to the main goal, which should be to prevent further environmental damage.

#### Overview

The discussion so far has identified significant, widespread, institutional underperformance and nonperformance across the GMB. But it is important to keep this matter in context. Institutional underperformance in GMB countries is a matter of choice. Each country is a "hard"—that is, authoritarian—state that regularly demonstrates its capacity to determine what can and cannot be done. Each country closely controls domestic political activity and the media and rigidly enforces internal security. Each country also spends lavishly on its public sector and subsidizes loss-making enterprises that serve their leaders' purposes. The inability of GMB countries to manage the basin's natural resources effectively, in large part, has been deliberate.

# SECTION 4: CHINA'S ROLE IN TRANSBOUNDARY RESOURCE MANAGEMENT?

China could help improve transboundary resource management across the GMB. It is the main upstream country and a major trading partner for all GMB countries and, in view of its national income and material wealth, it can support large-scale regional investments. As the regional hegemon and principal LMC partner, it has the institutional means to provide the necessary leadership.

.....

<sup>116</sup> First introduced by the King Bhumibol Adulyadej of Thailand in 1974, the idea of "sufficiency economy" was formalized in 1997 and incorporated in the 9th National Economic and Social Development Plan, 2001–2006 (Mongsawad 2010).

How might China induce GMB countries to follow its lead? First, information on all LMC activities will need to be made available. It is unreasonable to expect GMB countries to accept fully that the LMC is "a new sub-regional cooperation mechanism with unique features" serving their interests unless this can be publicly verified.<sup>117</sup> Opening up would involve regularly publishing electronic and print information on all LMC activities, along with their financing, who is implementing them and how, their timing, indicators for monitoring and evaluation, and how they contribute to LMC development goals.<sup>118</sup>

Second, local consultation and local participation should feature in all LMC activities. The current "leader-driven," bilateral, government-to-government, project-based model used by the LMC neglects these dimensions. China's insistence on doing *for* rather than doing *with* its GMB partners creates aid dependence, leaves undeveloped local capacities to operate efficiently and maintain effectively the capital and facilities created by LMC activities undeveloped, and potentially creates a level of recipient-country debt that cannot be serviced.

Third, China needs to move beyond its obsession with cooperation as the principal means of LMC engagement. The overemphasis on cooperation reinforces perceptions that its main function is to compel compliance.

Fourth, China should work to ensure that all GMB countries, itself included, account fully for the social costs of all LMC activities. The environmental deterioration throughout the GMB is direct evidence that exploiting natural resources was never a cheap way to grow and develop. Since, as noted earlier, China already has several efforts to appropriately value ecosystem services,<sup>119</sup> broadening the program across the GMB should be straightforward. A further advantage is that it would reinforce efforts to strengthen "green growth" and promote environmental sustainability.<sup>120</sup>

120 WWF (2014) and Walker (2019) specifically analsyze how the GMB and the BRI can be made "greener."

<sup>.....</sup> 

<sup>117</sup> TBNRM is relevant here as well. Van der Linde et al. (2001, p. 106) noted that "here is no blueprint for TBNRM" although they noted: "Communication is essential across the borders, within countries, within and across levels, and across institutional and technical sectors. This includes the sharing of information in a transparent and timely way" (p. 109).

Again, TBNRM provides useful lessons. Its practitioners emphasize the need for open, accountable, transparent procedures, regular monitoring and reporting, and widespread availability of information (van der Linde et al. 2001; Ch. 4; van der Linde 2013; Kaua 2015, p. 52; Okonkwo 2017).

<sup>119</sup> IUCN 2018, pp. 7, 12–22; Zhu and Ouyang 2019.

How would these changes improve transboundary resource management? Opening up informs all parties of the status and trends in natural-resource stocks and the flow of ecosystem services and how current and projected LMC projects affect them. Broader local consultation and participation would more closely match LMC activities to the needs and capacities of the communities being assisted. With improved information flows, the progress toward the LMC's development goals can be monitored and adjustments made, if needed. At a minimum, these changes would encourage all participants to accommodate differences in attitudes toward the transformation and use of natural resources, to accept opposing perspectives regarding their value and contribution to national income and welfare, and to consider disparate opinions regarding who gains and who loses and who should be compensated by whom, and how.

Another advantage of opening up and local participation is that LMC activities would be subjected to public scrutiny.<sup>121</sup> This may not reduce development disparities across the GMB, but it would prompt progress in that direction.<sup>122</sup> There are some obvious examples. The China-Laos high-speed rail referred to earlier is one. Hydro dams in Laos regularly displace thousands of households without adequate compensation while simultaneously exporting the low-priced electricity generated by the dams to boost the lifestyles of residents in Bangkok and elsewhere.<sup>123</sup> Government-to-government project selection in Thailand resulted in the high-speed rail link from China to Bangkok bypassing the country's northeast, which is already systematically deprived of development opportunities due to local political factors.<sup>124</sup> By contrast, local resistance is making it difficult for Vietnam's leaders to expand coal-fired generation

123 Sayatham and Sudhardiman 2015.

<sup>.....</sup> 

<sup>121</sup> Junyi Zhang 2016. Zhang Chao (2018) noted that the lack of transparency in China's foreign assistance raises the risk that its programs and their objectives will be misinterpreted both at home and abroad. Opening-up offers a chance to counteract this tendency, particularly between China and its GMB partners.

<sup>122</sup> Authoritarian governments are reluctant to open up, but the LMC's goals will not be met unless restrictions are eased so that information can be shared.

<sup>124</sup> National budgets provide evidence. A 2012 World Bank public financial management review of Thailand showed that the greater Bangkok area, which had 17 percent of the country's population and generated 26 percent of its GDP, was allocated 72 percent of the national budget. By contrast, Northeast Thailand, which has 34 percent of the country's population and generated 11.5 percent of its GDP, received 6 percent of the national budget (WB 2012a, p. 7).

capacity.<sup>125</sup> These and other examples show that if the LMC is to achieve its goals of reducing the development gaps, enhancing citizen welfare, and promoting social and economic development, the potential outcomes of its activities should be closely and critically scrutinized.<sup>126</sup>

What is the likelihood that China will make these changes? Until President Xi's April 2019 speech to the Second Belt and Road Forum, most observers would have argued that China had no incentive to modify its approach.<sup>127</sup> Up to that point, all high-level LMC communiques were reporting major progress and consistently adding more projects to the agenda.<sup>128</sup> It is inconceivable that any senior Chinese official would have advocated revising what is seen officially and advertised widely as a winning strategy. President Xi's pronouncements, however, may indicate some possibility for flexibility. His speech highlighted the need for transparency, opening up, avoiding debt traps, improving governance (especially "global governance"), and promoting innovation. He also spoke of the "need" for China to "be guided by the principle of extensive consultation, joint contribution, and shared benefits"; "pursue open, green, and clean cooperation"; "pursue high-standard cooperation to improve people's lives and promote sustainable development"; "take a people-centered approach, giv[ing] priority to poverty alleviation and job creation"; "ensure the commercial and fiscal sustainability of all projects so that they will achieve the intended goals as planned"; "make economic globalization more open, inclusive, balanced, and beneficial to all"; and "work harder to ensure the implementation of opening-up-related policies." He also said China would "adopt widely accepted rules and standards and encourage participating companies to follow general international rules and standards in project development, operation, procurement and tendering and bidding."129

The speech outlined an expansive agenda that, if implemented, would represent a fundamental shift in China's approach to BRI activities and presumably to the LMC

<sup>.</sup> 

<sup>125</sup> Mekong Eye 2016; TTN 2016; Le Quynh 2019; Khanh and Gloystein 2019.

<sup>126</sup> The development gap will not be reduced quickly. The study of GMS energy prospects noted "development gaps remain very large" (ADB 2013a, p. 88, Table 30).

<sup>127</sup> This was the conclusion of the version of this paper presented to the May 2019 workshop at Beijing Normual University.

<sup>128</sup> China, 2017, 2018; Lifang 2018; Li Keqiang 2018; Xiang Bo 2018.

<sup>129</sup> Xi 2019.

as well. If it comes, change is likely to be slow. It will take time to assess the impact of the President's announcements. Chinese officials recognize the bureaucratic risks involved and are likely to remain cautious, especially in the short-term. Thus far, the LMC has been administered so that GMB countries cooperate in ways that promote China's agenda. Nevertheless, some outside of China may not feel so constrained. The issues identified—recurrent costs, raising water efficiency and quality, and appropriate resource valuation—are critical to national development, essential for improved transboundary resource management and, thus far, largely unaddressed. Officials from other GMB countries just may take President Xi at his word and act accordingly.

#### SECTION 5: CONCLUDING COMMENTS

China could readily lead an effort to improve transboundary natural resource management across the GMB, although it is not clear that it will. At present, the LMC is serving its "leader-guided" purposes and GMB governments continue to support the LMC agenda. GMB countries are likely to require more time to determine whether government-to-government bilateral project selection and implementation is closing the GMB's development gap and serving other purposes such as regional development.

If China's leaders decide to reorient the LMC's administration, there is plenty to do. This essay has identified areas critical to improving transboundary resource management. Moreover, even if senior GMB officials are reluctant to shift direction, some useful experimentation may be possible at lower levels.

Numerous groups and individuals in the GMB, both inside and outside of government, have made noteworthy progress on topics related to improving transboundary resource management, and with the appropriate encouragement they may directly address the questions raised in this paper.

Four issues stand out.

• The recurrent-cost implications of the surge in BRI investment: Understanding the recurrent-cost implications will require a full accounting of the scale, scope, and timing of the investment expenditure being made to meet the

LMC's goals. It will also require detailed collaborative understanding of each country's budget allocations for operations and maintenance and how they are financed.

- Overcoming the institutional fragmentation that blocks transboundary water governance: Rationalizing the basin's approach to water governance would enable GMB countries to increase the effective supply of water through selective fees while simultaneously raising water quality.
- The collective need for GMB countries to appropriately value *all* of their productive resources and use those valuations in selecting future projects and programs: The appropriate valuation of all productive resources would prevent the massive social waste associated with continued environmental degradation. Furthermore, this approach would ensure that transboundary externalities are no longer overlooked, helping stimulate the collaboration required to enhance *regional* development.
- The collaboration required to formulate transboundary responses to climate change: A further advantage to appropriate valuation is that it would induce GMB countries to begin addressing climate change and dealing collectively with its consequences.

In principle, the LMC provides an institutional framework through which these issues could be addressed. That has not happened so far because of China's focus on using the LMC to drive its bilateral agendas with each GMB country. This situation need not persist. President Xi's references at the second Belt and Road Forum to opening-up, accountability, participation, consultation, people-centred approaches, green cooperation, and global partnership, among others, suggest that changes are possible. Other GMB countries should welcome this shift and, within the limits of their interests and capacities, use it to help meet the LMC's development goals.

### ANNEX I: THE LMC AND PUBLIC POLICY

Cooperation is a central feature of both the Sanya Declaration and the LMC Plan of Action. The LMC presumes that by cooperating, GMB countries will stimulate broad-based development. In their high-level meetings, GMB leaders have agreed to cooperate, but since they do not define the term, they have leeway to decide how cooperation fits their agenda or where their agenda is consistent with cooperating.<sup>130</sup>

In principle and in practice, cooperation cannot cope with the LMC's complexities. First, each country will have to decide what it means to cooperate and how they will cooperate, on what issues, and over what time periods, as well as the procedures through which cooperative activities will be derived and modified. Second, even when they do cooperate the outcomes and consequences will remain unclear. Game theorists remind us that cooperators always have mixed motives.<sup>131</sup> Goals and preferences, perceptions of gains and losses, willingness to bear risks, the time frame involved, and acceptable responses are never completely aligned and may even conflict.<sup>132</sup> Third, none of the GMB countries share the same worldview; relations among them at best have a checkered history, contentious issues continue to emerge, and each of them has competing interests. Offsetting these factors is that each country is tethered in multiple ways to the Lancang-Mekong River, and each has openly asserted that it values peaceful coexistence. Within the limits of their interests and mutual advantage, all of them appear willing to work together to develop the GMB.<sup>133</sup>

.....

- 132 Illustrated by Thomas Schelling's 1960 question: When two dynamite trucks meet on a road built for one, who backs up?
- 133 Whether the GMB countries embrace the notion of a community of common destiny (Mardell 2017; Zhang Denghua 2018, Truong-Minh Vu and Mayer 2018) or, as the LMC Plan of Action puts it, a "Community of Shared Future Peace and Prosperity," has yet to be demonstrated (Liangyu 2018). One difficulty is that there is little clarity on what "shared community" means in practice (EIAS 2018; Tan Kuan Hian 2019; Zeinullayev 2019).

<sup>130</sup> These qualifications highlight the point emphasized by Yeophantong (2017, p. 169) that compliance does not imply consent. She specifically criticized how cooperation features in GMB water governance. She aptly reminds us that "lack of contestation does not always indicate cooperation or concord."

<sup>131</sup> Schelling 1960, 1978; Aumann 1998; Maskin 2016. Scholars who have addressed GMB issues from a game-theoretic perspective include Belay et al. 2010; Singer (2015), Kim and Dinar (2015), Heydarian (2018) and Middleton (2018).

Fourth, sovereign nations typically welcome cooperation with their neighbors, but it is rarely adequate to sustain their ongoing relationships. All GMB governments will trade off the relative advantages of their prospective mutual engagement, reconcile divergent perceptions of the goals being sought and the risks involved, jointly negotiate how much and in what form they will contribute to the effort, and, in light of the progress they each measure, continue existing activities and selectively modify (or abandon) them.

The implication is that although cooperation has been an important motivator for devising the LMC, it will not be critical to its long-term achievements. This conclusion is unlikely to be accepted by LMC advocates, who will refer to the progress made so far through multilevel cooperation and, as noted earlier, will note that all GMB governments have regularly asserted their commitment to cooperate. Both are true, but beside the point. Government policy is not what is stated in action plans, strategic agendas, formal declarations, or legal instruments. It is reflected in behavior—what is done and left undone, what is funded and left unfunded, and where, how, and from whom revenue is derived. It is this behavior that reveals the government's priorities and defines its agenda.<sup>134</sup>

The LMC's emphasis on cooperation systematically ignores how governments act. The decisions that underpin their actions are motivated by questions like: Who gains and who loses from the activities chosen? Do winners compensate losers and by how much, over what period, and starting when? And how are public-private interactions negotiated and mediated, and in whose interests? None of these matters can be resolved by cooperation. Answers emerge as the respective governments act (or choose not to act) in accordance with their interests, goals, preferences, and capacities.

This point undermines the idea in the LMC Plan of Action that implementation will occur—as noted earlier—according to the principles of consensus, equality, voluntarism, and so on. In practice, LMC implementation will involve negotiation, conciliation, compromise, reciprocation, compensation, adjudication, concession, mutual

The emphasis on action is deliberate. It is consistent with a wide range of formal definitions. For example, Webster's New Collegiate Dictionary (1975, p. 890) defines policy as "a definite course or method of action selected from among alternatives," while dictionary.com defines it as "a definite course of action adopted for the sake of expediency, facility etc."

acceptability and, where appropriate, rejection. For GMB countries to make substantial, sustained progress toward the LMC's development goals, they will need to openly identify, debate, and analyze the relative benefits and costs associated with each LMC activity and, if possible, resolve any difficulties that may arise.

A fundamental problem with the LMC's overemphasis on cooperation is that it assumes GMB countries have deep reservoirs of mutual trust. Numerous examples cited earlier demonstrate that this is not the case. Successful implementation of the LMC agenda will require significant efforts to rebuild and sustain trust.<sup>135</sup> A useful start would be for GMB countries to agree on mechanisms to foster mutually beneficial interaction when cooperation proves inadequate. Efforts should be made to demonstrate to all GMB stakeholders that the LMC can help close the development gap. Relevant actions would be the assurance that all LMC-supported activities will be transparently and accountably designed and selected, that they provide value for money, and that they're being sustainably financed. Procedures to openly and fairly resolve technical and other differences among GMB partners will be needed as well. Finally, the performance of LMC projects will need to be monitored, clearly reported, and widely reviewed so that GMB partners and other relevant stakeholders can examine the outcomes, learn from them, and use the experience to guide future actions.

For its part, China can begin building trust by actively demonstrating that the activities it promotes through the LMC are consistent with its domestic policies such as "green growth" and environmental preservation.<sup>136</sup> This is currently contradicted by its support for coal-fired power plants in Cambodia and Vietnam.<sup>137</sup> The points about openness, transparency, and appropriate monitoring are obvious.

In the absence of a marked reduction in the LMC's reliance on cooperation, the arrangement faces two potentially unattractive futures. One is that the enthusiasm associated with the LMC's launch and the initial burst of activities will fade, and

<sup>.....</sup> 

<sup>135</sup> Lu Guangsheng 2016; Truong-Minh and Mayer 2018. The instrumental role of trust has been widely examined (Coleman 1990; Fukuyama 1995). Critical elements in fostering trust are predictability, reciprocity, transparency, and measured engagement (Yamagishi et al. 2005).

<sup>136</sup> Domestically, China is being guided by "the concept of innovative, coordinated, green, open and shared development" (Xinhua 2017b). So far, these grand gaols have not carried over to guide the operations of Chinese SOEs which are building LMC-sponsored infrastructure throughout the GMB.

<sup>137</sup> Cronin 2014; TTN 2016; Klare 2016; Cunningham 2017; Kotani 2018.

partners' attention will drift elsewhere. LMB countries already participate in several groupings (GMS, MRC, ASEAN, the Lower Mekong Initiative) that actively promote regional development.<sup>138</sup> The other scenario is that the partners remain engaged but in ways that involve, at most, relatively minor adjustments in their existing activities to align them with the LMC agenda as they interpret it. Such "muddling through" is the all-too-familiar bureaucratic response when operational requirements become too disruptive or the outcome is judged to be not worth the political capital expended.<sup>139</sup>

138 This may already be happening. Chandran (2019) cited a Center for Strategic and International Studies brief that inquired, "Will... [BRI] spending help people who need it most? Will it go into viable projects or white elephants? Will it help or hurt climate change? Will it create or destroy value?"

.....

<sup>&</sup>quot;Muddling through" refers to bureaucratic decision making in which governments, ministries, or departments do not act decisively to resolve an issue (Lindblom 1959). Solutions or outcomes are produced (or "ground out") by the existing institutional rules and relationships. Under these circumstances, policymaking becomes "a process of successive approximations to some desired objectives in which what is desired itself continues to change under reconsideration" (p. 86).

# ANNEX II: FOOD SECURITY, IRRIGATION, AND FLOOD CONTROL IN VIETNAM<sup>140</sup>

Decisions taken decades ago continue to affect agriculture and the environment in Mekong Delta in Vietnam. The government's quest for food security, irrigation expansion, and flood control illustrates how adverse effects can accumulate.

Vietnam's "rice first" policy was launched in the 1970s in response to widespread food deprivation following the end of the Vietnam War.<sup>141</sup> To boost national rice production, farmers in the Upper Mekong Delta were encouraged to plant three crops of rice a year rather than the traditional one or two. Despite repeated studies showing that the third crop loses money, farmers remain locked in. Communal irrigation schedules and their small plots prevent farmers from shifting out of rice or leaving their land fallow.

Triple-cropping rice reduces soil fertility and intensifies pressure from pests and diseases. Farmers respond by applying more agrochemicals. The Mekong Delta has no dedicated drainage, so irrigation water flows over farmers' fields and returns to the supply canal. This cascading water use increases the concentration of chemicals as the water moves down the Delta, making it progressively less suited for horticulture crops and aquaculture. Producers compensate by using groundwater. Extraction rates are now so pronounced that land subsidence is accelerating. The result is a relative sea-level rise that, when combined with declining dry-season river flows,<sup>142</sup> increases saline intrusion. Farmers, in turn, raise the rate of groundwater use.

These interconnected effects exacerbate the delta's water quality and environmental problems. Unremediated, they will continue the self-reinforcing adverse spiral of environmental degradation, increasing water shortages, and declining water quality.

<sup>140</sup> Example taken from Le and McPherson (2019).

<sup>141</sup> Ho and McPherson 2010; Le and McPherson 2015.

<sup>142</sup> Vietnamese officials regularly attribute the decline in dry-season flows to the operation of Chinese Lancang River dams. An inventory (S,s) or "bathtub" model will demonstrate that this has the story backwards. Chinese dams impound flood season flows which are then released during the dry season to generate electricity. This behavior boosts dry-season flows through the delta.

These problems are compounded by three factors. The persistence of a policy that equates food security with the supply of rice;<sup>143</sup> the development of irrigation in the delta for water supply only rather than as an integrated water supply and drainage system; and flood control levies and dikes that accelerate peak flood season water downriver leaving less water upriver to sustain dry-season flows.

A fourth factor is ineffectual environmental management. Vietnam's Ministry of Natural Resources and Environment (MONRE), which is responsible for implementing environmental law, lacks the human capacity and institutional stature to require farmers to reduce their use of agrochemicals.<sup>144</sup>

Viewed separately, each decision—achieving food security, expanding water supply, and flood control—could be readily justified, but officials have been slow to recognize their adverse cumulative impacts and take action to remedy them.

143 A World Bank study noted the high opportunity cost of continuing public irrigation investment to boost rice production (WB/VNM 2017, p. 30).

.....

<sup>144</sup> MONRE also lacks the capacity and institutional stature to prevent other agencies, particularly State-owned entities and other ministries, from polluting.

CHINA'S ROLE IN PROMOTING TRANSBOUNDARY RESOURCE MANAGEMENT IN THE GREATER MEKONG BASIN (GMB)

## REFERENCES

| ADB    | Asian Development Bank   |
|--------|--|
| СССРС  | Central Committee of the Communist Party of China                        |
| CEN    | China Economic Net   |
| EIAS   | European Institute for Asian Studies                                     |
| ESCAP  | Economic and Social Commission for Asia and the Pacific (United Nations) |
| FAO    | Food and Agriculture Organization of the United Nations                  |
| GWP    | Global Water Partnership   |
| ICEM   | International Centre for Environmental Management                        |
| IR     | International Rivers   |
| IUCN   | International Union for the Conservation of Nature                       |
| MEA    | Millennium Ecosystem Management  |
| MEP    | Ministry of Environmental Protection                                     |
| MRC    | Mekong River Commission  |
| NESDB  | National Economic and Social Development Board of Thailand               |
| OECD   | Organization for Economic Cooperation and Development                    |
| SIWI   | Stockholm International Water Institute                                  |
| TTN    | Tuoi Tre News  |
| UNDP   | United Nations Development Program                                       |
| VOA    | Voice of America   |
| WB     | World Bank   |
| WB/VNM | World Bank/Government of Vietnam   |
| WB/MPI | World Bank/Ministry of Planning and Investment (Vietnam)                 |
| WCS    | World Conservation Union   |
| WWF    | World Wildlife Fund  |
|        |  |

- Abbott, James, et al. (2007) "Rivers as resources, rivers as borders: community and transboundary management of fisheries in the Upper Zambezi River floodplains," *The Canadian Geographer*, vol. 51, no. 3, pp. 280–302.
- Ahuja, Ambiha (2010) "Q+A: Is China to blame for the Mekong drying up?" *Reuters*, World News, April 6.
- Albert, Eleanor, and Beina Xu (2016) "China's Environmental Crisis," Council on Foreign Relations, *Backgrounder*, January 18.

- Allan, J. David, et al. (2005) "Overfishing of Inland Waters," *Bioscience*, vol. 55, no. 12, pp. 1041–1051.
- Antikainen, Juhani, Rita Gebert, and Ulf Moller (2011) "Review of the Greater Mekong Sub-Region Regional Power Trade," Final Report, SIDA Review 2011:9, Swedish International Development Cooperation Agency, Stockholm.
- Aoyama, Rumi (2016) "One Belt, One Road: China's New Global Strategy," *Journal of Contemporary East Asian Studies*, vol. 5, no. 2, pp. 3–22.
- Asian Development Bank (2005) *Connecting East Asia: A New Framework for Infrastructure,* Tokyo: Asian Development Bank, Japan Bank for International Cooperation, and Washington, D.C.: The World Bank.
- Asian Development Bank (2010) "Strategy and Action Plan for the Greater Mekong Subregion East-West Economic Corridor," Mandaluyong City: Asian Development Bank.
- Asian Development Bank (2011) *The Greater Mekong Subregion Economic Cooperation Program Strategic Framework*, Metro Manila: Asian Development Bank.
- Asian Development Bank (2012a) *Infrastructure for Supporting Inclusive Growth and Poverty Reduction in Asia*, Mandaluyong City: Asian Development Bank.
- Asian Development Bank (2012b) *Greater Mekong Subregion Power Trade and Interconnection:* 2 Decades of Cooperation, Mandaluyong City: Asian Development Bank.
- Asian Development Bank (2013a) Assessment of the Greater Mekong Subregion Energy Sector Development: Progress, Prospects, and Regional Investment Priorities, Mandaluyong City: Asian Development Bank.
- Asian Development Bank (2013b) *Viet Nam: Environment and Climate Change Assessment*, Metro Manilla: Asian Development Bank.
- Asian Development Bank (2016a) "Review of Configuration of the Greater Mekong Subregion Economic Corridors," Metro Manila: Asian Development Bank.
- Asian Development Bank (2016b) "Summary of Proceedings: 19th Meeting of the GMS Regional Power Trade Coordination Committee (RPTCC-19)," Mandaluyong City: Asian Development Bank.
- Asian Development Bank (2017) "Asia's Infrastructure Needs Exceed \$1.7 Trillion Per Year: Double Previous Estimate," Asian Development Bank News Release, February 28.
- Asian Development Bank (2018) *Greater Mekong Subregion: Twenty-Five Years of Partnership,* Metro Manila: Asian Development Bank.
- Asian Development Bank (2019) "Infrastructure," *OpenDevelopment Mekong*, February 18.

- Aumann, Robert J. (1998) "game theory," in John Eatwell, Murray Milgate, and Peter Newman (eds.), *The New Palgrave: A Dictionary of Economics*, Volume 2, London: The Macmillan Press Ltd, pp. 460–479.
- Bache Hanne et al. (2012) *Transboundary River Basin Management: Addressing Water, Energy and Food Security,* Vientiane: Mekong River Commission.
- Bai Tiantian (2017) "Lancang-Mekong Cooperation secretariat opens," *Global Times,* November 3.
- Banomyong, Ruth (2008) "Logistics Development in the Greater Mekong Subregion," Power-Point presentation, Center for Logistics Research, Thammasat University, Bangkok.
- Banomyong, Ruth (2010) "Benchmarking economic corridors logistics performance: a GMS border crossing observation." *World Customs Journal*, vol. 4, no. 1, pp. 37–46.
- Banomyong Ruth (2013) "The Greater Mekong Sub-region of Southeast Asia: Improving Logistics Connectivity," in Bookbinder, James H. (ed.) Handbook of Global Logistics: International Series in Operations Research & Management Science, vol 181. New York: Springer.
- Bastiaanssen, Wim, and Xueliang Cai (2018) "Variability of Field Scale Water Productivity in Different Crops as a Basis for Policy Making," Delft: International Institute for Hydraulic and Environmental Engineering (IHE), October 2.
- Baten, Mohammed Abdul, and Rashed Al Mahmud Titumir (2016) "Environmental challenges of trans-boundary water resources management: the case of Bangladesh," *Sustainable Water Resources Management*, vol. 2, pp. 13–27.
- Belay, Alebel Abebe, et al. (2010) "The Challenges of Integrated Management of Mekong River
  Basin in Terms of People's Livelihoods," *Journal of Water Resource and Protection*, vol. 2, pp. 61–68.
- Berdik, Chris (2014) "Of Fish, Monsoons and the Future," New York Times, op-ed, June 9.
- Bernstein, Richard (2017) "China's Mekong Plans Threaten Disaster for Countries Downstream," *Foreign Policy*, September 17.
- Bloomberg Editorial Board (2018) "A New Direction for China's Belt and Road," *Bloomberg*, November 1.
- Brady, David, and Michael Spence (2010) *Leadership and Growth Commission on Growth and Development*, Washington, D.C.: The World Bank.
- Breu, Marco, Richard Dobbs, and Jaana Remes (2012) "Taking Vietnam's economy to the next level," McKinsey & Company.

- Bright, Gareth (2018) "Gallery: Intensive Agriculture in Vietnam," ChinaDialogue.net, February 22.
- Brilingaite, Vilte (2017) "China's Transboundary River Governance: The Case of the Lancang-Mekong River," Center for East and South-East Asian Studies, Lund University.
- Bruce-Lockhart, Anna (2017) "China's \$900 billion New Silk Road. What you need to know," *World Economic Forum*, June 26.
- Buchanan, James M. (1998) "opportunity cost," in John Eatwell, Murray Milgate, and Peter Newman (eds.), *The New Palgrave: A Dictionary of Economics*, vol. 3, New York: The Stockton Press, pp. 718–721.

Buchanan, James M. (2003) "What Is Public Choice Theory?" *Imprimis*, vol. 32, no. 3.

- Butler, Rhett (2013) "Mekong region has lost a third of its forests in 30 years, may lose another third by 2030," mongabay.com, May 3.
- Cambodia, Royal Government of (2018) *Rectangular Strategy for Growth, Equity and Efficiency: Building the Foundation Toward Realizing the Cambodia Vision 2050, Phase IV,* Phnom Penh: Royal Government of Cambodia of the Sixth Legislature of the National Assembly.
- Campbell, Lauren, et al. (2015) "Environmental impact assessment: Theory, practice and its implications for the Mekong hydropower debate," *International Journal of Water Governance*, no. 5, pp. 93–116.
- Carter, Becky (2017) "A literature review on China's aid," Knowledge, evidence and learning for Development (K4D), Institute of Development Studies, September 7.
- Central Committee of the Communist Party of China (2016) "The 13th Five-Year Plan for Social and Economic Development of the People's Republic of China (2016–2020)," Central Compilation & Translation Press, Beijing.
- Chandran, Nyshka (2018) "China can make its Belt and Road project more successful if it taps locals, experts say," *China Politics,* September 14.
- Chandran, Nyshka (2019) "Japan, not China, may be winning Asia's infrastructure investment contest," CNBC.com, January 23.
- Chanthavilay, Minaphone, et al. (2017) "Assessment of Wastewater Treatment in Developing Countries: Current Practices in Vientiane Capital, Laos," *International Journal of Advanced Research (IJAR)*, vol. 4, no. 4, pp. 335–344.
- Chea, Ratha, Gael Grenouillet, and Sovan Lek (2016) "Evidence of Water Quality Degradation in Lower Mekong Basin Revealed by Self-Organizing Map," *PLOS one*, vol. 11, no.1, January, p. 19.
- Chen Jia (2019) "Fund mulled to support Belt, Road projects," ChinaDaily, March 18, pp. 1, 3.

- Chen-Sheng Hong and Oliver Johnson (2018) "Mapping potential climate and development impacts of China's Belt and Road Initiative: a participatory approach," Stockholm Environment Institute, Bangkok.
- China, People's Republic of (2017) "Joint Press Communique of the Third Lancang-Mekong Cooperation (LMC) Foreign Ministers' Meeting," Ministry of Foreign Affairs, Beijing, December 16.
- China, People's Republic of (2018) "Li Keqiang Visits Lancang-Mekong Cooperation Exhibition Together with Leaders of Five Mekong River Countries," Phnom Penh, Ministry of Foreign Affairs, January 11.
- ChinaDaily (2018) "Five-Year Plan of Action on Lancang-Mekong Cooperation (2018–2022)," *ChinaDaily*, January 11.
- China Economic Net (2013) "China pledges further efforts to boost western growth," *China Economic Net*, December 21.
- China Economic Net (2019) "China's 'Go West' Strategy," China Economic Net, January 25.
- China Economic Net (2015) "New 'Water Ten Plan' to Safeguard China's Water," *China Water Risk*, Hong Kong, April 16.
- China, National Bureau of Statistics (2019) *China Statistical Yearbook 2018*, Beijing: China Statistical Press.
- Chong Koh Ping (2017) "China is one of the top foreign aid providers but only some projects have strong economic impact: Study," *The Straits Times*, October 11.
- Clark, Pilita (2014) "Troubled waters: the Mekong River crisis," *The Financial Times*, July 18.
- Coleman, James S. (1990) *Foundations of Social Theory*, Cambridge, MA: The Belknap Press of Harvard University Press.
- Connor, Logan (2019) "Cambodia's Tonle Sap Lake: where fishermen have no fish and no hope," *South China Morning Post*, January 20.
- Copper, John F. (2016) *China's Foreign Aid and Investment Diplomacy Volume III: Strategy Beyond Asia and Challenges to the United States and the International Order*, London: Palgrave MacMillan.
- Costanza, Robert, et al. (1997) "The value of the world's ecosystem services and natural capital," *Nature*, vol. 387, May 15, pp. 253–260.
- Costanza, Robert, et al. (2014) "Changes in the global value of ecosystem services," Global Environmental Change, vol. 26, pp. 152–158.
- Cronin, Richard (2014) "Finding Solutions to Equitable Hydropower Development Planning in the Lower Mekong Basin," Stimson Center, Washington D.C.

Cunningham, Edward (2017) "Chinese Markets and Understanding Xi's Priorities," Annual Report 2017, Greentech Capital Advisors.

Daming He et al. (2014) "China's transboundary waters: new paradigms for water and ecological security through applied ecology," *The Journal of Applied Ecology*, vol. 51, no. 5, October, pp. 1159–1168.

Dang Kieu Nhan (2013) "Water use efficiency, competition and conflict in the Mekong Delta, Vietnam," PowerPoint presentation, East Asia and Pacific–Agriculture and Rural Development Community of Practice Meeting, Can Tho University, June 10.

- Dang Kieu Nhan et al. (2011) "Water Use, Competition and Potential Conflict in Vietnam's Mekong Delta," World Bank Assessment Report (Project #7155810), Mekong River Development Institute, Can Tho University, November 27.
- Dara, Mech, and Kuoch Masy (2018) "Ministry gets told off over bad road quality," *Phnom Penh Post,* January 7.
- Davies, George R. (2013) "Appraising Weak and Strong Sustainability: Searching for a Middle Ground," *Consilience: The Journal of Sustainable Development*, vol. 10, no. 1, pp. 111–124.
- Deane, Phyllis (1965) The First Industrial Revolution, Cambridge: Cambridge University Press.
- Devlaeminck, David J. (2018) "Timeline of the Lancang-Mekong Cooperation (LMC) Mechanism," *ResearchGate*, April.
- Dixon, John A., and Kirk Hamilton (1996) "Expanding the Measure of Wealth," *Finance & Development*, vol. 33, no. 4, pp. 15–18.
- Dobson, Andrew (1996) "Environmental Sustainabilities: an analysis and a typology," *Environmental Politics*, vol. 5, no. 3, pp. 401–428.

DTiNews (2017) "Air quality worsens in Hanoi," *Vietnam Insider*, July 28.

- Duggan, Jennifer (2014) "China's polluters to face large fines under law change," *Guardian*, April 25.
- Economic and Social Commission for Asia and the Pacific (2018) "Report of the Mekong River Commission," Economic and Social Commission for Asia and the Pacific (United Nations), Seventy-fourth session, Bangkok, May 11–16.

*Economist, The* (2009) "China's "Going Out" Strategy," Free Exchange, *The Economist*, July 21. *Economist, The* (2011) "Trying to pull together: The Chinese in Africa," *The Economist*, April 20.

*Economist, The* (2015) "Chinese Investment in Africa: Not as easy as it looks," *The Economist,* Middle East and Africa, November 21.

Economist, The (2016) "Requiem for a river," The Economist, February 13.

Economist, The (2017) "What is China's belt and road initiative?" The Economist, May 15.

*Economist, The* (2018) "Is China investing too much in infrastructure?" *The Economist*, June 1. European Institute for Asian Studies (2018) "China's Belt and Road Initiative is neither a strat-

egy nor a vision. It is a process," *EU-Asia at a Glance*, European Institute for Asian Studies, February 4.

- Eyler, Brian, and Courtney Weatherby (2019) "Sustaining the heartbeat of the Mekong Basin," *China Dialogue*, March 11.
- Fang Wan (2018) "How 'unconditional' is China's foreign aid?" Deutche Welle (DW), May 15.
- Ferchen, Matt (2018) "Where's the 'development' in China's global development finance?" ChinaDialogue.net, September 1.
- French, Howard W. (2010) "The Next Empire," *The Atlantic*, May 28.
- Fukuyama, Francis (1995) *Trust, The Social Virtues and the Creation of Prosperity,* London, Hamish Hamilton.
- Geall, Sam (2017) "Clear waters and green mountains: Will Xi Jinping take the lead on Climate Change?" Lowy Institute, Sydney.
- Global Water Partnership (2015) "China's waste resources management challenge; The 'three red lines,'" Global Water Partnership, Stockholm.
- Granick, David (1957) "Economic Development and Productivity Analysis: The Case of Soviet Metalworking," *Quarterly Journal of Economics*, vol. 71, no. 2, May, pp. 205–233.
- Gray, Clive S., and Andre Martens (1982) "The Political Economy of the 'Recurrent Cost Problem' in the West African Sahel," *World Development*, vol. 11, no. 2, pp. 101–117.
- Greer, Tanner (2018) "One Belt, One Road, One Big Mistake," *Foreign Policy*, December 6.
- Griffiths, James (2018) "Belt and Road: Are wheels coming off China's megaproject?" CNN, December 31.
- Hall, David S., and Kanokwan Manorom (2015) "Decision-Making in the Mekong: Science, Nature and Society," *Global Asia*, vol. 10, no. 1, Spring, p. 26.
- Hart, Barry T., Michael J. Jones, and Gabrielle Pistone (2001) "Transboundary Water Quality Issues in the Mekong River Basin," mekonginfo.org.
- Hatton, Celia (2017) "China's secret aid empire uncovered," BBC News, October 11.
- Hediger, Werner (2008) "Weak and Strong Sustainability, Environmental Conservation and Economic Growth," *Natural Resource Modeling*, vol. 19, no. 3, September, pp. 359–394.
- Heller, Peter S. (1979) "The Underfinancing of Recurrent Development Costs," *Finance & Development*, vol. 16, no. 1, pp. 38–41.
- Heydarian, Richard (2018) "Proactive Diplomacy' in Mekong River dispute only way to resolve brewing conflict," *South China Morning Post*, January 14.

- Ho Dang Hoa and Malcolm McPherson (2010) "Policy Brief: Food Security and Land Policy in Vietnam," *Sage Journals*, November 20.
- Hoogzaad, Jelmer, et al. (2017) *Circular Economy Strategies for Lao PDR*, Vientiane: Ministry of Energy and Mines and United Nations Development Program, September 11.
- CWR (2018) "No Water, No Growth Does Asia have enough water to develop?" China Water Risk, Hong Kong, September 21.
- International Centre for Environmental Management/International Union for the Conservation of Nature (2013) "PAD2: Protected Areas, Development and Climate Change in the Greater Mekong Subregion," Hanoi: International Centre for Environmental Management, and Gland, Switzerland: International Union for the Conservation of Nature.
- International Rivers (2013) "Lancang River Dams: Threaten the Flow of the Lower Mekong," International Rivers, Berkeley, CA, August.
- International Union for the Conservation of Nature (2018) IUCN 2017, Annual Report 2017, Bangkok: International Union for Conservation of Nature Regional Office.
- International Union for the Conservation of Nature/Wildlife Conservation Society/World Wildlife Fund (2017) "Consuming the Future: The real status of biodiversity in Lao PDR," Village Focus International, Vientiane.
- Janekarnkij, Penporn, and Orn-uma Polpanich (2014) "Valuing Ecosystem Services in the Mekong Region," in Louis Lebel et al. *Climate Risks, Regional Integration and Sustainability in the Mekong Region,* Kuala Lumpur: Strategic Information and Research Development Center.
- Jin Kai (2014) "China's Two-Front Diplomacy: 'Go West' and 'Leap East'" *The Diplomat*, September 12.
- Johnson, D. Gale (2000) "Population, Food, and Knowledge," *The American Economic Review*, vol. 90, no. 1, pp. 1–14.
- Junyi Zhang (2016) "Chinese foreign assistance, explained," Brookings.edu, July 19.
- Kacungira, Nancy (2017) "Will Kenya get value for money from its new railway?" *BBC Africa*, June 8.
- Kaua, Caxton Gitonga (2015) "Transboundary National Resource Management: Rationale, Challenges and Way Forward," Academia.edu.
- Khanh Vu and Flenning Gloystein (2019) "In Vietnam's booming energy sector coal reigns, but renewables play catch-up," *Reuters*, May 24.
- Kim Hang Pham Do and Ariel Dinar (2015) "Issue Linkage as a Mechanism for Conflict Management in the Mekong Basin," in Ariel Dinar and Yacov Tsur, *Management of Transboundary Water Resources Under Scarcity*, World Scientific Publishing Company.

- Kitano, Naohiro (2016) "Estimating China's Foreign Aid II: 2014 Update," Tokyo: Japan International Cooperation Agency.
- Klare, Michael T. (2016) "Coal: investment—and resistance," *MAC: Mines and Communities*, June 20.
- Kobayashi, Kiyoshi, et al. (2017) *Economic Integration and Regional Development*, London: Routledge.
- Kolinjivadi, Vijay K., and Terry Sunderland (2012) "A Review of Two Payment Schemes for Watershed Services from China and Vietnam: the Interface of Government Control and PES Theory," *Ecology and Society*, vol. 17, no. 4, essay 10.
- Kotani, Hiroshi (2018) "Coal backlash creates energy dilemma in Southeast Asia," *Nikkei Asian Review*, January 19.
- Kummu, Matti, and Juha Sarkkula (2008) "Impact of the Mekong River Flow Alteration on the Tonle Sap Flood Pulse," *Ambio*, vol. 37, no. 3, pp. 185–192.
- Kuo, Lily, and Niko Kommenda (2018) "What is China's *Belt* and *Road* Initiative?" *The Guardian*, July 30.
- Kwong Man-ki (2015) "China's east-west income gap widens," *South China Morning Post*, May 28.
- Kynge, James (2018) "China's Belt and Road projects drive overseas debt fears," *The Financial Times*, August 7.
- Lao PDR (2011) "The Seventh Five-Year National Socio-Economic Development Plan (2011– 2015)," Lao People's Democratic Republic Ministry of Planning and Investment, Vientiane, October 7.
- Lao PDR (2016) "8th Five-Year National Socio-Economic Development Plan (2016–2020)," Lao People's Democratic Republic Ministry of Planning and Investment, Vientiane, April.
- Le Quang Minh (2001) "Environmental Governance: A *Mekong Delta* Case Study with Downstream Perspectives," College of Technology, Cantho University.
- Le Quynh (2019) "Vietnamese provinces say 'no'" to coal plants—government and industry still want more," MekongEye.com, March 7.
- Le Thi Quynh Tram and Malcolm F. McPherson (2015) "The Future of Agriculture in the Lower Mekong Basin," Ho Chi Minh City: Fulbright School of Public Policy and Management and Cambridge, MA: Ash Center for Democratic Governance and Innovation, Harvard Kennedy School, Cambridge, December.
- Le Thi Quynh Tram and Malcolm F. McPherson (2019) "Vietnam's Environmental Challenges and Opportunities: The Special Role of Agriculture," Fulbright University of Vietnam, Ho Chi Minh City.

48

- Lebel, Louis, et al. (eds.) (2014), *Climate Risks, Regional Integration and Sustainability in the Mekong Region*, Kuala Lumpur: Strategic Information and Research Development Center (SIRD).
- Lebel, Louis, Suchada Wattana, and Pwin Talerngsri (2015) "Assessments of ecosystem services and human well-being in Thailand build and create demand for coproductive capacity," *Ecology and Society*, vol. 20, no. 1.
- Leibold, James (2014) "Is China's 'Go West' Strategy Doomed?" The National Interest, May 28.
- Liangyu (ed.) (2018) "Spotlight: China makes five proposals on securing shared future for Lancang-Mekong nations," XinhuaNet.com, January 11.
- Lifang (2018) "China pledges another 6 bln USD loans for Lancang-Mekong cooperation," XinhuaNet.com, January 11.
- Li Hongmei (2013) "New energy Silk Road to boost West China's development," *China Economic Net*, November 27.
- Li Keqiang (2018) "Building bridges of friendship and cooperation in the Lancang-Mekong region and between China and Cambodia," *Khmer Times*, January 9.
- Lindblom, Charles E. (1959) "The Science of 'Muddling Through," *Public Administration Review*, vol. 19, no. 2, pp. 79–88.
- Liu Jiaxin (2018) "12-day training program on Lancang-Mekong Cooperation opens in Beijing," CGTN.com, June 29.

Lu Guangsheng (2016) "China Seeks to Improve Mekong Sub-Regional Cooperation: Causes and Policies," S. Rajaratnam School of International Studies, Nanyang Technological University.

- Mardell, Jacob (2017) "The 'Community of Common Destiny' in Xi Jingping's New Era," *The Diplomat*, October 25.
- Maskin, Eric (2016) "How Can Cooperative Game Theory Be Made More Relevant to Economics?:
  An Open Problem," in J. F. Nash Jr. and Themistocles M. Rassias (eds.), *Open Problems in Mathematics*, Geneva: Springer International Publishing, pp. 347–350.
- Matson, Pamela, William C. Clark, and Krister Andersson (2016) *Pursuing Sustainability: A Guide to the Science and Practice*, Princeton, N.J.: Princeton University Press.
- McPherson, Malcolm F. (2012) "Land Policy in Vietnam: Challenges and Prospects for Constructive Change," *Journal of Macromarketing*, vol. 32, no. 1, pp. 137–146.
- Mekong Eye (2016) "Coal Power on the Rise: Mekong Region Digs In," MekongEye.com, February 15.
- Mekong River Commission (2004) "Proceedings Policy Dialogue on Watershed Management in the Lower Mekong Basin," Vientiane: Mekong River Commission.

Mekong River Commission (2005) "Overview of the Hydrology of the Mekong Basin," Vientiane: Mekong River Commission.

Mekong River Commission (2009) "Adaptation to climate change in the countries of the Lower Mekong Basin," Vientiane: Mekong River Commission.

- Mekong River Commission (2012) "Basin-wide collaboration in the agriculture and irrigation sub-sectors towards the development and food security in the Lower Mekong Basin," Vientiane: Mekong River Commission.
- Mekong River Commission (2015) *Mekong River Commission: 20 Years of Cooperation*, Vientiane: Mekong River Commission.
- Menon, Jayant, and Peter Warr (2013) "The Lao Economy: Capitalizing on Natural Resource Exports," Mandaluyong City: Asian Development Bank.

Meyn, Colin, and Mech Dara (2017) "City of garbage," Southeast Asia Globe, September 14.

- Middleton, Carl (2018) "Can Chinese 'reciprocity' protect the Mekong?" ChinaDialogue.net, November 12.
- Middleton, Carl, and Jeremy Allouche (2016) "Watershed or Powershed? Critical Hydropolitics, China and the 'Lancang-Mekong Cooperation Framework,'" *The International Spectator*, vol. 51, no. 3, pp. 110–117.
- Millennium Ecosystem Assessment (2005) "Living Beyond Our Means: Natural Assets and Human Well-Being," Washington D.C.: Millennium Ecosystem Assessment.
- Millward, James A. (2018) "Is China a Colonial Power?" New York Times, May 4.
- Ministry of Environmental Protection (2017) "The Belt and Road Ecological and Environmental Cooperation Plan," Beijing: Ministry of Environmental Protection.
- Mongsawad, Prasopchoke (2010) "The Philosophy of the Sufficiency Economy: A Contribution to the Theory of Development," *Asia-Pacific Development Journal*, vol. 17, no. 1, pp. 123–143.

Moxley, Mitch (2010) "China's Poor West Gets an Uncertain Push," Inter Press Service, July 21.

- Murphy, Tom (2017) "Find Out Some (But Not All) The Secrets of China's Foreign Aid," *NPR*, October 31.
- Myanmar, Government of (2018) "Myanmar Sustainable Development Plan (2018–2030)" Yangon: Office of the State Counsellor.

Nash, Paul (2012) "China's 'Going Out' Strategy," *Diplomatic Courier*, May 10.

National Economic and Social Development Board (2017) "The Twelfth National Economic and Social Development Plan (2017–2021)," Bangkok: Office of the National Economic and Social Development Board, Office of the Prime Minister.

Nguyen Huu Dung and Agnes Rola (1999) "Agro-chemicals, Productivity, and Health in Vietnam," Los Baños, Laguna, Philippines: Economics and Environmental Program for South East Asia.

Nguyen Khac Giang (2018) "China is making Mekong friends," eastasiaforum.org, May 19.

- Nguyen Tin Hong (2017) "An Overview of Agricultural Pollution in Vietnam: The Crops Sector," Washington, D.C.: The World Bank Group.
- Nguyen Van Cong (2017) "An Overview of Agricultural Pollution in Vietnam: The Aquaculture Sector," Washington, D.C.: The World Bank Group.
- Nguyen Van Kien and Huynh Ngoc Duc (2015) "Conserving the benefits of floating rice in Vietnam," *Better Ways*, mekongcommons.org.

Oirere, Shem (2016) "Kenya readies first standard-gauge line," RailJournal.com, June 16.

- Okonkwo, Theodore (2017) "Management of transboundary natural resources," *Journal of Law and Conflict Resolution*, vol. 9, no. 4, November, pp. 42–52.
- Organization for Economic Cooperation and Development (2013) "Structural Policy Country Notes: Lao PDR," extract from *Economic Outlook for Southeast Asia, China and India* 2014: Beyond the Middle-Income Trap, Paris: Organization for Economic Cooperation and Development.
- Organization for Economic Cooperation and Development (2015) "OECD Principles on Water Governance," Paris: Organization for Economic Cooperation and Development.
- Organization for Economic Cooperation and Development (2018) *Implementation of the OECD Principles on Water Governance Indicator Framework and Evaluation Practices*, Paris: OECD Publishing.
- Page, John (1994) "The East Asian Miracle: Four Lessons for Development Policy," Stanley Fischer and Julio M. Rotemberg (eds.), NBER Macroeconomics Annual 1994, Cambridge: National Bureau for Economic Research, pp. 219–282.
- Perdiguero, Alfredo (2016) "Economic Corridor Development: The Greater Mekong Subregion Experience," PowerPoint presentation, G-20 Global Infrastructure Connectivity Forum, Singapore, April 27.
- Rasanen, Timo A. (2017) "New study shows significance of China's dams on Mekong," TheThirdPole.net, February 6.
- Roberts, Amada M. (2003) "The Role of Transboundary Natural Resource Management in Peace Building Across International Borders," Convention on Biological Diversity.
- Roberts, David, and Jalel Sager (2016) "Recharging Asia's Batteries: What Obama Should Do in Laos," *Foreign Affairs*, vol. 95, no. 5, September 1.

- Saunders, Jade (2014) "Illegal Logging and Related Trade: The Response in Lao PDR," London: Chatham House.
- Sayatham, Mayvong, and Diana Sudhardiman (2015) "Hydropower resettlement and livelihood adaptation: The Nam Mang 3 project in Laos," *Water Resources and Rural Development,* vol. 5, pp. 17–30.

Schelling, Thomas C. (1960) *The Strategy of Conflict*, Cambridge, Mass.: Harvard University Press. Schelling, Thomas C. (1978) *Micromotives and Macrobehavior*, New York: W.W. Norton.

- Schmittmann, Jochen, David Corvino, and Misturu Katagiri (2017) "Environment and Climate Change," Vietnam: Selected Issues, International Monetary Fund IMF Country Report No. 17/191.
- Schneider, Leander (2010) "Schneider on Monson, 'Africa's Freedom Railway: How a Chinese Development Project Changed Lives and Livelihoods in Tanzania," networks.h-net.org.
- Seiff, Abby (2019) "At a Cambodian Lake, a Climate Crisis Unfolds," *New York Times*, September 30.
- Serey, Sok (2012) "PM Wants Better Roads," Radio Free Asia, October 11.
- Singer, Emily (2015) "Game Theory Calls Cooperation into Question," quantamagazine.org, February 23.
- Singh, Naina (2016) "Yunnan: China's Bridge to South and Southeast Asia," *The Diplomat,* August 26.
- Small, Andrew (2018) "The Backlash to Belt and Road," Foreign Affairs, February 16.
- Solanes, Miguel, and Fernando Gonzalez-Villarreal (1999) "The Dublin Principles for Water as Reflected in a Comparative Assessment of Institutional and Legal Arrangements for Integrated Water Resources Management," Santiago, Chile: Global Water Partnership Technical Advisory Committee, Santiago Chile.
- Soong, Jennifer (2006) "Soil fertility and changes in fertilizer use for intensive rice cultivation in Red River Delta and Mekong Delta of Vietnam," DigitalCollections.sit.edu, 340.
- Sopheap, Kry (2016) "Poor Waste Management Threatens Phnom Penh," San Francisco: The Asia Foundation, July 27.
- Standaert, Michael (2017) "As It Looks to Go Green, China Keeps a Tight Lid on Dissent," e360.Yale.edu, November 2.
- State Council (2011) *China's Foreign Aid*, Beijing: The State Council of the People's Republic of China.
- State Council (2014) *White Paper on China's Foreign Aid*, Beijing: Xinhua/Information Office of State Council.

State Council (2015) "Full Text: Integrated Reform Plan for Promoting Ecological Progress," Beijing: The State Council of the People's Republic of China, September 22.

Stiglitz, Joseph E., Amartya Sen, and Jean-Paul Fitoussi (the Sarkozy Commission) (2009) "Report by the Commission on the Measure of Economic Performance and Social Progress," at https://ec.europa.eu/eurostat/documents/118025/118123/Fitoussi+Commission+report.

- Strange, Austin M. (2018) "China's Foreign Aid Beyond the Headlines," Medium.com/ Fairbank-Center, February 12.
- Sullivan, Michael (2018) "China Reshapes the Vital Mekong River to Power Its Expansion," *NPR*, October 6.
- Sungsup Ra and Zhigang Li (2018) "Closing the Financing Gap in Asian Infrastructure," Asian Development Bank South Asia, *Working Paper Series*, no. 57.
- Tan Kuan Hian (2019) "Tighter Belts, Fewer Roads: Challenges to China's Belt and Road Initiative," Masters Thesis, Graduate School of Arts and Sciences, Harvard University.
- Tarp, Finn, et al. (2007) "Accounting for Vietnam's Wealth: The Role of Natural Resources," ideas.repec.org, no. 29391.
- Terra Daily (2013) "Deforestation threatens Mekong region," TerraDaily.com, Staff Writers, May 2.
- Thomas, Brent (2018) "Laos gambles on becoming 'battery of South East Asia,'" *Southeast Asia Globe*, August 3.
- Truong Minh Vu and Maximilian Mayer (2018) "Hydropower infrastructure and regional order making in the Sub-Mekong Basin," *Revista Brasileira de Politica Internacional*, vol. 61, no. 1.
- Tung Xuan Dinh (2017) "An Overview of Agricultural Pollution in Vietnam: The Livestock Sector," World Bank Regional Agricultural Pollution Study 2017, Washington, D.C.: The World Bank Group.
- *Tuoi Tre News* (2016) "Coal-fired power plants threaten Vietnam deltas," MekongEye.com, *Tuoi Tre News*, October 5.
- Tweed, David (2018) "China's New Silk Road," *Bloomberg*, August 23.
- Un, Kheang, and Sokbunthoeun So (2009) "Politics of Natural Resource Use in Cambodia," *Asian Affairs: An American Review,* vol. 36, no. 3, pp. 123–138.
- United Nations (2015) *Transforming our world: The 2030 Agenda for Sustainable Development,* Report A/RES/70/1, New York: United Nations.
- United Nations Development Program/Stockholm International Water Institute (2016) "Water Governance in perspective: Water Governance Facility 10 years, 2005–2015," United

Nations Development Program and Stockholm International Water Institute, New York and Stockholm.

- Urban, Frauke, Giusseppina Siciliano, and Johan Nordensvard (2018) "China's dam-builders: their role in transboundary river management in South-East Asia," *International Journal of Water Resources Development*, vol. 34, no. 5, pp. 447–770.
- van der Linde, Harry, et al. (2001) Beyond Boundaries: Transboundary Natural Resource Management in Sub-Saharan Africa, Washington D.C.: Biodiversity Support Program.
- van der Linde, Harry (2013) "When and How to Go Transboundary Effective Strategies for Transboundary Natural Resource Management," PowerPoint presentation, iSlideDocs.com, January 24.
- van Dijk, Meine Pieter (2011) "A Different Development Model in China's Western and Eastern Provinces," *Modern Economy*, vol. 2, pp. 757–768.
- van Mead, Nick (2018) "China in Africa: win-win development, or new colonialism?" *The Guard-ian*, July 31.
- Vardon, Michael (2014) "Australia's Accounts Show Value of Natural Capital is over AUD\$4.8 trillion," Wealth Accounting and the Valuation of Ecosystem Services (WAVES), The World Bank, wavespartnership.org.
- Vietnam, Socialist Republic of (2011) "Vietnam's Socio-Economic Development Strategy for the Period of 2011–2020," Prime Minister Decision No. 432/QD-TTg, Hanoi, April 12.

Vietnam, Socialist Republic of (2012a) "National Strategy on Environment Protection to 2020 with Visions to 2030," Prime Minister Decision No. 1216/QD-TTg, Hanoi, September 5.

- Vietnam, Socialist Republic of (2012b) "Viet Nam Sustainable Development Strategy for 2011– 2020," Prime Minister Decision No. 432/QD-TTg, Hanoi, April 12.
- Vietnam, Socialist Republic of (2013) "Approving the Project 'Agricultural Restructuring Towards Raising Added Values and Sustainable Development," Prime Minister Decision No. 899/ QD-TTg, Hanoi, June 10.
- Vietnam, Socialist Republic of (2014) "Law on Environmental Protection," No. 55/2014/QH13, The National Assembly, Hanoi, June 23.
- Vietnam, Socialist Republic of (2016) *The Five-Year Socio-Economic Development Plan 2016–2020*, Hanoi: Government of Vietnam.
- Voice of America (2018) "Pollution Clouds Vietnam's Rapid Economic Growth," Voice of America, East Asia, VOANews.com, May 15.
- von Ciriacy-Wantrup, Siegfried (1952) *Resource Conservation: Economics and Policies,* Berkeley: University of California Press.

von Ciriacy-Wantrup, Siegfried (1961) "Conservation and Resource Programming," *Land Economics*, vol. 37, no. 2, May, pp. 105–111.

VNNet (2017) "Pollution getting worse in Vietnam's cities," english.vietnamnet.vn, July 18.

Walker, Gabriel (2019) "Green Finance for Chinese Overseas Investment," Masters Thesis, John F. Kennedy School of Government, Harvard University.

Wang Hongying (2016) "A Deeper Look at China's "Going Out" Policy, Center for International Governance and Innovation, CIGI Commentary, March.

Wang-Kaeding, Heidi (2018) "What does Xi Jinping's New Phrase 'Ecological Civilization' Mean?" *The Diplomat*, March 6.

Wang Ruka (2017) "China's Addiction to Unproductive Infrastructure, "*Chinese Politics*, www .uscnpm.org/blog/2017, July 7.

Wang Yan (2018) "Can the countries of the Mekong pioneer a new model of cooperation?" TheThirdPole.net, March 15.

Wang Zhihe, He Huili, and Fan Meijan (2014) "The Role of Ecological Marxism and Constructive Post-Modernism—Beyond the Predicament of Legislation," *Monthly Review*, vol. 66, no. 6.

Weatherby, Courtney, and Brian Eyler (2016) "Letters from the Mekong: Mekong Power Shift— Emerging Trends in the GMS Power Sector," Washington, D.C.: The Stimson Center.

Wharton (2013) "The Economics of China's Pollution Problem," Knowledge.Wharton.UPenn .edu, May 1.

World Bank (1993) *The East Asian Miracle: Economic Growth and Public Policy,* Washington D.C.: Oxford University Press for the World Bank.

World Bank (1994) *World Development Report 1994: Infrastructure for Development*, New York: Oxford University Press for the World Bank.

World Bank (2008) *The Growth Report: Strategies for Sustained Growth and Inclusive Development*, Washington D.C.: The World Bank.

World Bank (2009) *World Development Report 2009, Reshaping Economic Geography,* Washington D.C.: The World Bank.

World Bank (2012a) *Improving Service Delivery: Thailand: Public Finance Management Report, Overview,* Bangkok: The World Bank, Report Number 67486-TH.

World Bank (2012b) "Factoring Nature into National Wealth Accounting," worldbank.org, April 12.

World Bank (2015) "The Wealth Accounting and Valuation of Ecosystem Services Global Partnership Program: Frequently Asked Questions," worldbank.org, June 15.

World Bank (2016) "Transforming Vietnamese Agriculture: Gaining More From Less," worldbank .org, April 29.

- World Bank (2017a) "An Overview of Agricultural Pollution in Vietnam The Crops Sector," Washington, D.C.: The World Bank Group.
- World Bank (2017b) Cambodia: Sustaining Strong Growth for the Benefit of All: A Systematic Country Diagnostic, Washington D.C.: World Bank Group, Report No. 115181-KH.
- World Bank (2018) "Belt and Road Initiative," worldbank.org, March 29.
- World Bank (2018) "Infrastructure gaps vary across East Asia and the Pacific and between Cities and Rural Areas," worldbank.org, March 28.
- World Bank/Ministry of Planning and Investment (2016) *Vietnam 2035: Toward Prosperity, Creativity, Equity, and Democracy*, Washington D.C.: The World Bank and Vietnam: Ministry of Planning and Investment.
- World Bank/Government of Vietnam (2017) "Vietnam Public Expenditure Review Fiscal Policies towards Sustainability, Efficiency, and Equity," A Joint Review, Washington, D.C.: The World Bank and the Government of Vietnam.
- World Wildlife Fund (2014) "Green Growth in the Greater Mekong Subregion," Gland, Switzerland: World Wildlife Fund for Nature.
- Xi Jinping (2017) "Secure a Decisive Victory in Building a Moderately Prosperous Society in All Respects and Strive for the Great Success of Socialism with Chinese Characteristics for a New Era," english.qstheory.cn, October 18.
- Xi Jinping (2018) "Speech delivered by Xi Jinping at the first session of the 13th NPC," *China-Daily*, March 21.
- Xi Jinping (2019) "Working Together to Deliver a Brighter Future for Belt and Road Cooperation," fmprc.gov.cn, April 26.
- Xiang Bo (2018) "Mekong River countries can become major cooperation partners in Belt and Road Initiative: Chinese state councilor," XinhuaNet.com, March 31.
- Xinhua (2011) "China's Foreign Aid," Xinhua News Agency, April 21.
- Xinhua (2016) "Sanya Declaration of the First Lancang-Mekong Cooperation (LMC) Leader's Meeting," March 23.
- Xinhua (2017a) "Xi leads ecological civilization," *China Daily*, March 22.
- Xinhua (2017b) "Xi elaborates on inspiration behind Belt and Road Initiative," XinhuaNet.com, May 15.
- Xinhua (2019) "Main section of China-Laos Railway bridge over Mekong River completed," XinhuaNet.com, August 15.
- Yamagishi, Toshio, et al. (2005) "Separating Trust from Cooperation in a Dynamic Relationship," *Rationality and Society*, vol. 17, no. 3, pp. 275–308.

- Yana Jin, Hendrik Andersson, and Shiqiu Zhang (2016) "Air Pollution Control Policies in China: A Retrospective and Prospects," *International Journal of Environmental Research and Public Health*, vol. 13, no. 12.
- Yanzhong Huang (2018) "Why China's Good Environmental Policies Have Gone Wrong," *New York Times,* January 14.
- Yeophantong, Pichamon (2016) "China's Dam Diplomacy in the Mekong Region—Three Game Changers," D. Blake and L. Robins (eds.), *Water Governance Dynamics in the Mekong Region*, Selangor, Malaysia: Strategic Information and Research Development Center Press.
- Yeophantong, Pichamon (2017) "River activism, policy entrepreneurship and transboundary water disputes in Asia," *Water International*, vol. 42, no. 2, pp. 163–186.
- Yun Sun (2014) "China's Aid to Africa: Monster or Messiah?" Brookings.edu, February 7.
- Zeinullayev, Tlek (2019) "Neo-mercantilism with Chinese Characteristics," Masters Thesis, John F. Kennedy School of Government, Harvard University.
- Zeitoun, Mark, Marisa Goulden, and David Tickner (2013) "Current and future challenges facing transboundary river basin management," ResearchGate.net.
- Zhang Chao (2018) "Why Is Transparency So Important for China's Development Assistance?" *The Diplomat*, November 10.
- Zhang Denghua (2018) "The concept of 'Community of Common Destiny' in China's Diplomacy: Meaning, Motives and Implications," *Asia & The Pacific Policy Studies*, vol. 5, no. 2, pp. 196–207.
- Zhang Jiaqi (2018) "2 new environmental laws to go into effect in 2018," china.org.cn, January 1.
- Zhang Hongzhou and Li Minjiang (2018) "Thirsty China and its transboundary waters," Hongzhou Zhang and Minjiang Li (eds.), *China and Transboundary Water Politics in Asia*, New York: Routledge.
- Zhang Yue (2017) "Lancang-Mekong Cooperation to grow," *China Daily,* December 16.
- Zhong Yong (2018) "LMC Water Cooperation," PowerPoint presentation, MRCsummit.org, April 2.
- Zhou, Laura (2018) "Are Chinese-funded dams on the Mekong River washing away Cambodian livelihoods?" *South China Morning Post*, July 3.
- Zhou, Margaret (2016) "China and the Mekong Delta: Water Savior or Water Tyrant?" *Earth Island Journal*, April 15.
- Zhou Shixin (2018) "A Strategic Vision of the Lancang-Mekong Cooperation Mechanism," *Crossroads: South East Asia*, January at http://siis.org.cn.

- Zhu Chunquan and Ouyang Zhiyun (2019) "Gross Ecosystem Product," PowerPoint presentation, seea.un.org, February 13.
- Zhu Feng (2013) "New Silk Road Economic Belt—Spearhead of China's Westward Strategy," *China Today*, December 3.
- Zhu Shaoping and Kong Xiaoli (2016) "Problem of Water Management System in Our Country and Reform Train of Thought," *Advances in Environmental Protection*, vol. 6, no. 3, pp. 39–45.





A publication of the

Ash Center for Democratic Governance and Innovation Harvard Kennedy School 79 John F. Kennedy Street Cambridge, MA 02138

617-495-0557